Rice Energy Inc. Form 4 June 05, 2015

FORM 4

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

STATEMENT OF CHANGES IN BENEFICIAL OWNERSHIP OF

SECURITIES

OMB Number:

3235-0287

OMB APPROVAL

Expires:

January 31, 2005

0.5

Estimated average burden hours per

response...

if no longer subject to Section 16. Form 4 or Form 5 obligations

Check this box

Filed pursuant to Section 16(a) of the Securities Exchange Act of 1934, Section 17(a) of the Public Utility Holding Company Act of 1935 or Section

may continue. 30(h) of the Investment Company Act of 1940 See Instruction

1(b).

(Print or Type Responses)

1. Name and Address of Reporting Person * 5. Relationship of Reporting Person(s) to 2. Issuer Name and Ticker or Trading VAGT ROBERT F Issuer Symbol Rice Energy Inc. [RICE] (Check all applicable) (First) (Middle) (Last) 3. Date of Earliest Transaction (Month/Day/Year) X_ Director 10% Owner Other (specify Officer (give title 400 WOODCLIFF DRIVE 06/03/2015 below) (Street) 4. If Amendment, Date Original 6. Individual or Joint/Group Filing(Check Filed(Month/Day/Year) Applicable Line) _X_ Form filed by One Reporting Person Form filed by More than One Reporting CANONSBURG, PA 15317 Person

(City) (State) (Zip) Table I - Non-Derivative Securities Acquired, Disposed of, or Beneficially Owned 1.Title of 2. Transaction Date 2A. Deemed 4. Securities Acquired 5. Amount of 6. Ownership 7. Nature of Security (Month/Day/Year) Execution Date, if Transaction(A) or Disposed of Securities Form: Direct Indirect (Instr. 3) Code (D) Beneficially (D) or Beneficial anv (Month/Day/Year) (Instr. 8) (Instr. 3, 4 and 5) Owned Indirect (I) Ownership Following (Instr. 4) (Instr. 4) Reported (A) Transaction(s) (Instr. 3 and 4) Code V Amount (D) Price Common 06/03/2015 $A^{(1)}$ 11,216 A \$0 37,085 D Stock

Reminder: Report on a separate line for each class of securities beneficially owned directly or indirectly.

Persons who respond to the collection of SEC 1474 information contained in this form are not (9-02)required to respond unless the form displays a currently valid OMB control number.

Table II - Derivative Securities Acquired, Disposed of, or Beneficially Owned (e.g., puts, calls, warrants, options, convertible securities)

1. Title of	2.	3. Transaction Date	3A. Deemed	4.	5.	6. Date Exerc	cisable and	7. Title and	8. Price of	9. Nu
Derivative	Conversion	(Month/Day/Year)	Execution Date, if	Transactio	orNumber	Expiration D	ate	Amount of	Derivative	Deriv
Security	or Exercise		any	Code	of	(Month/Day/	Year)	Underlying	Security	Secui
(Instr. 3)	Price of		(Month/Day/Year)	(Instr. 8)	Derivativ	e		Securities	(Instr. 5)	Bene
	Derivative		•		Securities	S		(Instr. 3 and 4)		Owne
	Security				Acquired			· · · · · · · · · · · · · · · · · · ·		Follo
	·				(A) or					Repo
					Disposed					Trans
					of (D)					(Instr
					(Instr. 3,					
					4, and 5)					
				Code V	(A) (D)	Date	Expiration	Title Amount		
						Exercisable	Date	or		
								Number		
								of		
								Shares		

Reporting Owners

Reporting Owner Name / Address	Relationships					
	Director	10% Owner	Officer	Other		
VAGT ROBERT F						
400 WOODCLIFF DRIVE	X					
CANONSBURG, PA 15317						

Signatures

/s/ Robert F. Vagt, by William E. Jordan, as Attorney-in-Fact

06/05/2015

**Signature of Reporting Person

Date

Explanation of Responses:

- * If the form is filed by more than one reporting person, see Instruction 4(b)(v).
- ** Intentional misstatements or omissions of facts constitute Federal Criminal Violations. See 18 U.S.C. 1001 and 15 U.S.C. 78ff(a).
- (1) Restricted stock unit award under the Issuer's Amended and Restated Rice Energy Inc. 2014 Long-Term Incentive Plan. The restricted stock units vest in a single installment on June 3, 2016.

Note: File three copies of this Form, one of which must be manually signed. If space is insufficient, *see* Instruction 6 for procedure. Potential persons who are to respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB number. ed to the Company, the Company believes that, except for one late filing by Sherman McCorkle disclosing one purchase of shares through Mr. McCorkle's 401(K) plan, there was compliance with all filing requirements of Section 16(a) applicable to directors and executive officers of the Company during fiscal year 2010. During fiscal year 2010, no person or entity beneficially owned more than 10 percent of the Company's Common Stock.

28

Reporting Owners 2

PROPOSAL II:

RATIFICATION OF THE APPOINTMENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

The Audit Committee of the Board of Directors has appointed KPMG LLP ("KPMG") to serve as the Company's independent registered public accounting firm for the fiscal year 2011 and the Board of Directors recommends that shareholders ratify such appointment at the Annual Meeting.

Action by the shareholders is not required by law in the appointment of an independent registered public accounting firm, but their appointment is submitted by the Audit Committee of the Board of Directors in order to give the shareholders a voice in the designation of auditors. If the resolution ratifying the appointment of KPMG as the Company's independent registered public accounting firm is rejected by the shareholders, then the Audit Committee may reconsider its choice of independent registered public accounting firms. Even if the resolution is approved, the Audit Committee in its discretion may direct the appointment of a different independent registered public accounting firm at any time during the year if it determines that such a change would be in the best interests of the Company and its shareholders. Proxies in the form solicited hereby that are properly submitted will be voted FOR the resolution unless otherwise instructed by the shareholder.

Representatives of KPMG are expected to attend the Annual Meeting. They will have an opportunity to make a statement if they desire to do so, and are expected to be available to respond to appropriate questions. Representatives of Deloitte & Touche LLP ("Deloitte"), EMCORE's prior independent accounting firm, are not expected to attend the Annual Meeting.

FISCAL 2010 & 2009 AUDITOR FEES AND SERVICES

KPMG LLP's ("KPMG") services as EMCORE's principal registered accountant commenced on March 5, 2010. The fees billed to the Company by KPMG in fiscal 2010 are as follows:

Audit fees \$372,902 (1)
Audit-related fees -Tax fees -All other fees --

29

Total

Fiscal 2010

\$372,902

⁽¹⁾ Represents fees for professional services rendered in connection with the integrated audit of our annual financial statements and reviews of our quarterly financial statements.

Table of Contents

Deloitte was the independent registered public accounting firm that audited EMCORE's financial statements for fiscal 2009 and was EMCORE's independent registered public accountant until March 4, 2010 for fiscal 2010. The aggregate fees billed by Deloitte in fiscal 2009 and 2010 are as follows:

	Fiscal 2010		Fiscal 2009(1)	
Audit fees	\$938,624	(2)	\$ 1,386,567	(2)
Audit-related fees (3)			4,950	(3)
Tax fees	61,329	(4)		
All other fees				
Total	\$999,953		\$ 1,391,516	

- (1) Fees reported for fiscal 2009 have been adjusted to represent fees actually billed in fiscal 2009, rather than fees billed in connection with services rendered for fiscal 2009.
- (2) Represents fees for professional services rendered in connection with the integrated audit of our annual financial statements and management's assessment of our internal control over financial reporting, reviews of our quarterly financial statements, other SEC filings, including registration statements, and correspondence with the SEC, and advice provided on accounting matters that arose in connection with audit services.
- (3) Represents fees for professional services related to our audits and reviews of our consolidated financial statements that are not considered audit fees including the audits of employee benefit plans during fiscal 2008 and other statutory or regulatory and SEC filings.
- (4) Represents fees for professional services related to a tax advice regarding our entry into a joint venture.

Audit Committee Pre-Approval Policies and Procedures

These services may include audit services, audit-related services, tax services and other services. The Audit Committee has adopted a policy for the pre-approval of services provided by the independent auditors. Under the policy, pre-approval is generally provided for up to one year and any pre-approval is detailed as to the particular service or category of services and is subject to a specific budget. In addition, the Audit Committee may also pre-approve particular services on a case-by-case basis. For each proposed service, the independent auditors are required to provide detailed back-up documentation at the time of approval. Pursuant to the Sarbanes-Oxley Act of 2002, all of the fees and services provided as noted in the table above were authorized and approved by the Audit Committee in compliance with the pre-approval policies and procedures described herein.

Changes in Registrant's Certifying Accountant

On March 3, 2010, the Audit Committee of the Board of Directors determined that it would be in the best interests of the Company to dismiss Deloitte as the Company's independent registered public accounting firm. Deloitte was notified of this action on March 4, 2010.

Deloitte's report on the Company's financial statements for the fiscal year ended September 30, 2009 did not contain any adverse opinion or disclaimer of opinion, nor was it qualified or modified as to uncertainty, audit scope, or any accounting principles. Deloitte's report on the Company's financial statements for the fiscal year ended September 30, 2008 expressed an unqualified opinion and included an explanatory paragraph relating to the Company's ability to continue as a going concern.

Table of Contents

During the Company's fiscal years ended September 30, 2009 and 2008 and through March 4, 2010, there were no disagreements with Deloitte on any matters of accounting principles or practices, financial statement disclosure, or auditing scope and procedures; which disagreements, if not resolved to the satisfaction of Deloitte, would have caused Deloitte to make reference to such disagreements in their reports on the Company's financial statements for such years. In addition, during the same period, there were no reportable events as described in Item 304(a)(1)(v) of Regulation S-K.

The Company provided Deloitte with a copy of the foregoing disclosures and requested Deloitte furnish the Company with a letter addressed to the SEC stating whether or not it agreed with the above statements. A copy of the letter from Deloitte, dated March 8, 2010, is attached as an exhibit to the Company's Current Report on Form 8-K filed with the SEC on March 9, 2010.

On March 5, 2010, the Company engaged KPMG as its independent registered public accounting firm for the fiscal year ended September 30, 2010, effective as of March 5, 2010. During the Company's two fiscal years ended September 30, 2009, and 2008 and the subsequent interim period through March 4, 2010, the Company did not consult with KPMG regarding any of the matters described in Item 304(a)(2)(i) and (ii) of Regulation S-K.

Table of Contents

REPORT OF THE AUDIT COMMITTEE

The following Report of the Audit Committee does not constitute soliciting material, and should not be deemed filed or incorporated by reference into any other Company filing under the Securities Act or the Exchange Act, except to the extent the Company specifically incorporates this Report of the Audit Committee by reference therein.

The Company has a separately-designated standing Audit Committee established in accordance with Section 3(a)(58)(A) of the Exchange Act. The Audit Committee currently consists of Messrs. Scott (chairman), Bogomolny, Gillen and McCorkle. Each member of the Audit Committee is currently an independent director within the meaning of Nasdaq Listing Rule 5605(a)(2). The Board of Directors has determined that Messrs. Scott and Gillen are each Audit Committee financial experts. The Audit Committee met 17 times in fiscal 2010. The Audit Committee performs the functions set forth in the EMCORE Corporation Audit Committee Charter, which has been adopted by the Board of Directors. The Audit Committee Charter is available in the Investor Relations section of our website (www.emcore.com).

The Audit Committee has reviewed and discussed the Company's audited financial statements for fiscal 2010 with management of the Company. The Audit Committee has discussed with the Company's independent registered public accounting firm the matters required to be discussed by Statement of Auditing Standards No. 114 Furthermore, the Audit Committee has reviewed management's assessment of the effectiveness of the Company's internal controls over financial reporting, and has reviewed the opinion of the Company's independent registered public accounting firm regarding such assessment and the effectiveness of the Company's internal controls over financial reporting.

The Audit Committee has received the written disclosures and the letter from the Company's independent registered public accounting firm required by applicable requirements of the Public Company Accounting Oversight Board regarding the independent registered public accounting firm's communications with the Audit Committee concerning independence, and has discussed with such accounting firm the independence of such accounting firm. Based on the foregoing review and discussions, the Audit Committee recommended to the Board of Directors that the Company's audited financial statements be included in the Company's Annual Report on Form 10-K for fiscal 2010, which was filed on January 10, 2011.

This report is submitted by the Audit Committee.

March 2, 2011

AUDIT COMMITTEE Charles T. Scott, Chairman Robert L. Bogomolny John Gillen Sherman McCorkle

RECOMMENDATION OF THE BOARD OF DIRECTORS

THE BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS A VOTE "FOR" THE RATIFICATION OF THE APPOINTMENT OF KPMG LLP AS THE COMPANY'S INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM UNDER PROPOSAL II.

Table of Contents

PROPOSAL III: APPROVAL OF A POSSIBLE REVERSE STOCK SPLIT IN THE RANGE FROM 2:1 TO 10:1, IN THE DISCRETION OF OUR BOARD OF DIRECTORS

Overview

The shareholders are being asked to approve a possible reverse stock split of our outstanding Common Stock in the range of from 2:1 to 10:1 (or more plainly stated, the range would be as low as from two existing shares to be exchanged for one new share, to as high as ten existing shares to be exchanged for one new share), as determined in the sole discretion of the Board of Directors. The Board of Directors has adopted a resolution (i) declaring the advisability of a possible reverse stock split in the range of 2:1 to 10:1, and a corresponding reduction in the number of authorized shares of the Company's Common Stock, subject to shareholder approval, (ii) in connection therewith, approving a form of amendment to our Restated Certificate of Incorporation to effect such a reverse stock split, subject to shareholder approval, and (iii) authorizing any other action the Board of Directors deems necessary to effect such a reverse stock split, without further approval or authorization of the Company's shareholders. If the proposed reverse stock split is approved, our Board of Directors would have the discretion to elect, as it determines to be in the best interests of the Company and its shareholders, to effect the reverse stock split at any exchange ratio within the range at any time before the date of our next annual meeting of shareholders. The Board of Directors may elect not to implement the approved reverse stock split at its sole discretion. The Board of Directors believes that approval of a proposal granting this discretion to the Board of Directors provides the Board of Directors with appropriate flexibility to achieve the purposes of the reverse stock split, if implemented, and to act in the best interests of the Company and its shareholders. The amendment to the Restated Certificate of Incorporation is attached as Exhibit A to this Proxy Statement.

The Purpose for Which the Board of Directors Would Effect the Reverse Stock Split

In order to attempt to proportionally raise the per share price of our Common Stock, the Board of Directors believes that it is in the best interests of our shareholders for the Board of Directors to obtain the authority to implement a reverse stock split. The Board of Directors believes that shareholder approval of a range of potential exchange ratios (rather than a single exchange ratio) provides the Board of Directors with the flexibility to achieve the desired results of the reverse stock split. If the shareholders approve this proposal, the Board of Directors would carry out a reverse stock split only upon the Board of Directors' determination that a reverse stock split would be in the best interests of the shareholders at that time. To accomplish the reverse stock split, we would file an amendment to the Restated Certificate of Incorporation with the New Jersey Secretary of State. The form of amendment to the Restated Certificate of Incorporation to accomplish the proposed reverse stock split is attached as Exhibit A to this Proxy Statement. The text of the amendment to the Restated Certificate of Incorporation is subject to modification to include such changes as the Board of Directors deems necessary and advisable to effect the reverse stock split, including the applicable ratio for the reverse stock split. If the Board of Directors elects to implement the reverse stock split, the number of issued and outstanding shares of our Common Stock would be reduced in accordance with the selected exchange ratio for the reverse stock split, and the number of authorized shares of the Common Stock would also be reduced in the same proportion. The number of authorized shares of Preferred Stock would remain unchanged. The reverse stock split would become effective upon filing the amendment to the Restated Certificate of Incorporation with the New Jersey Secretary of State. No further action on the part of shareholders would be required to either effect or abandon the reverse stock split. If the Board of Directors does not implement the reverse stock split prior to the date of our next annual meeting of shareholders, the authority granted in this proposal to implement the reverse stock split will terminate. The Board of Directors reserves its right to elect not to proceed and abandon the reverse stock split if it determines, in its sole discretion, that this proposal is no longer in the best interests of our shareholders.

Potential Effects of the Proposed Reverse Stock Split

The immediate effect of the reverse stock split would be to reduce the number of shares of the outstanding Common Stock and to increase the trading price of such Common Stock. However, the effect of any consummated reverse stock split upon the market price of the Common Stock cannot be predicted, and the history of reverse stock splits for companies in similar circumstances is that sometimes stock performance improves, but in many cases it does not. There can be no assurance that the trading price of the Common Stock after the reverse stock split will rise in proportion to the reduction in the number of shares of our Common Stock outstanding as a result of the reverse stock split or remain at an increased level for any period. Also, there is no assurance that a reverse stock split would not eventually lead to a decrease in the trading price of the Common Stock, that the trading price would remain above the thresholds required by the Nasdaq listing rules or that we will be able to continue to meet the other continued listing requirements of Nasdaq. The trading price of the Common Stock may change due to a variety of other factors, including our operating results, other factors related to our business and general market conditions. In addition, the liquidity of our Common Stock could be adversely affected by the reduced number of shares that would be issued and outstanding if the reverse stock split is effectuated.

Effects on Ownership by Individual Shareholders

If we implement the reverse stock split, the number of shares of our Common Stock held by each shareholder would be reduced by multiplying the number of shares held immediately before the reverse stock split by the selected exchange ratio, and then rounding up to the nearest whole share. The reverse stock split would not affect any shareholder's percentage ownership interest in the Company or proportionate voting power, except to the extent that interest in a fractional share would be rounded up to the nearest whole share.

Effect on Options, Warrants and Other Securities

In addition, all outstanding options, warrants and other securities entitling their holders to purchase shares of our Common Stock would be adjusted as a result of the reverse stock split, as required by the terms of these securities. In particular, proportionate adjustments will be made to the exercise price per share and the number of shares issuable upon the exercise of all outstanding options, entitling the holders to purchase shares of our Common Stock, which will result in approximately the same aggregate price being required to be paid for such options upon exercise immediately preceding the reverse stock split. Also, the number of shares reserved for issuance under any existing employee stock option plans would be reduced proportionally based on the selected exchange ratio of the reverse stock split. Our preferred stock would not be affected by the proposed reverse stock split.

Other Effects on Outstanding Shares

If the reverse stock split were implemented, the rights and preferences of the outstanding shares of the Common Stock would remain the same after the reverse stock split. Each share of Common Stock issued pursuant to the reverse stock split would be fully paid and non-assessable. The reverse stock split would result in some shareholders owning "oddlots" of less than 100 shares of Common Stock. Brokerage commissions and other costs of transactions in odd-lots are generally higher than the costs of transactions in "round-lots" of even multiples of 100 shares.

Effect on Authorized Shares of Common Stock

The reverse stock split, if implemented, would reduce the number of authorized shares of Common Stock in the same proportion as the selected exchange ratio for the reverse stock split (not accounting for any rounding of fractional shares as described below).

Procedure for Implementing the Proposed Reverse Stock Split and Exchange of Stock Certificates

If shareholders approve the proposed amendment to the Restated Certificate of Incorporation, the Board of Directors may elect whether or not to declare a reverse stock split at any time before our next annual meeting of shareholders. The reverse stock split would be implemented by filing the amendment to the Restated Certificate of Incorporation with the New Jersey Secretary of State, and the reverse stock split would become effective on the date the filing is accepted by the Secretary of State. As of the effective date of the reverse stock split, each certificate representing shares of our Common Stock before the reverse stock split would be deemed, for all corporate purposes, to evidence ownership of the reduced number of shares of our Common Stock resulting from the reverse stock split, except that holders who have not exchanged their certificates would not be entitled to receive any dividends or other distributions payable by us after the effective date until they surrender their old stock certificates for exchange. All shares, underlying options and warrants and other securities would also be automatically adjusted on the effective date. If we elect to exchange share certificates, our transfer agent would act as the exchange agent for purposes of implementing the exchange of stock certificates. In such event, as soon as practicable after the effective date, shareholders and holders of securities convertible into our Common Stock would be notified of the effectiveness of the reverse stock split. Shareholders of record would receive a letter of transmittal requesting them to surrender their stock certificates

for stock certificates reflecting the adjusted number of shares as a result of the reverse stock split. Persons who hold their shares in brokerage accounts or "street name" would not be required to take any further actions to effect the exchange of their certificates. No new certificates would be issued to a shareholder until such shareholder has surrendered the outstanding certificate(s) together with the properly completed and executed letter of transmittal to the exchange agent. Until surrender, each certificate representing shares before the reverse stock split would continue to be valid and would represent the adjusted number of shares based on the exchange ratio of the reverse stock split, rounded up to the nearest whole share. Shareholders should not destroy any stock certificate and should not submit any certificates until they receive a letter of transmittal.

Accounting Consequences

The no par value per share of our Common Stock would remain unchanged after the reverse stock split. The per share Common Stock net income or loss and net book value will be increased for current and comparative prior periods because there will be fewer shares of the Common Stock outstanding. We do not anticipate that any other accounting consequences would arise as a result of the reverse stock split.

Fractional Shares

We will not issue fractional shares in connection with the reverse stock split. In order to avoid the expense and inconvenience of issuing and transferring fractional shares of our Common Stock to shareholders who would otherwise be entitled to receive fractional shares of Common Stock following the reverse stock split, any fractional shares which result from the reverse stock split will be rounded up to the next whole share.

No Appraisal Rights

Under the New Jersey Business Corporation Act, shareholders are not entitled to appraisal rights with respect to the proposed amendment to the Restated Certificate of Incorporation to effect the reverse stock split.

United States Federal Income Tax Consequences of the Reverse Stock Split

The following is a summary of important U.S. tax considerations of the reverse stock split. It addresses only shareholders who hold the pre-reverse stock split shares and post-reverse stock split shares as capital assets. It does not purport to be complete and does not address shareholders subject to special rules, such as financial institutions, tax-exempt organizations, insurance companies, dealers in securities, mutual funds, foreign shareholders, shareholders who hold the pre-reverse stock split shares as part of a straddle, hedge, or conversion transaction, shareholders who are subject to the alternative minimum tax provisions of the Internal Revenue Code of 1986, as amended (the "Code"), and shareholders who acquired their pre-reverse stock split shares pursuant to the exercise of employee stock options or otherwise as compensation. This summary is based upon current law, which may change, possibly even retroactively. It does not address tax considerations under state, local, foreign, and other laws. Furthermore, we have not obtained a ruling from the Internal Revenue Service or an opinion of legal or tax counsel with respect to the consequences of the reverse stock split. Each shareholder is advised to consult a qualified tax advisor.

The proposed reverse stock split is intended to be treated as a "reorganization" within the meaning of Section 368 of the Code. Assuming the reverse stock split qualifies as a reorganization, a shareholder generally will not recognize gain or loss on the reverse stock split. The aggregate tax basis of the post–reverse stock split shares received will be equal to the aggregate tax basis of the pre–reverse stock split shares exchanged therefor (excluding any portion of the holder's basis allocated to fractional shares), and the holding period of the post–reverse stock split shares received will include the holding period of the pre–reverse stock split shares exchanged. The rounding up in respect of fractional shares will not result in a taxable event to a shareholder; however, there will be an adjustment to the shareholder's basis equal to the fractional share times the market value on the date of issuance. No gain or loss will be recognized by us as a result of the reverse stock split.

Required Vote

The affirmative vote of a majority of the votes represented in person or by proxy at the Annual Meeting is required for the adoption of the proposed amendment to the Restated Certificate of Incorporation to authorize us to effect a reverse stock split of our Common Stock in the range from 2:1 to 10:1, as determined in the sole discretion of the our Board of Directors. Proxies will be voted in accordance with the specifications marked thereon, and, if no specification is made, will be voted "FOR" the proposed reverse stock split.

RECOMMENDATION OF THE BOARD OF DIRECTORS

THE BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS THAT SHAREHOLDERS VOTE "FOR" THE AMENDMENT TO THE COMPANY'S RESTATED CERTIFICATE OF INCORPORATION TO EFFECT A REVERSE STOCK SPLIT IN THE RANGE FROM 2:1 TO 10:1, IN THE DISCRETION OF OUR BOARD OF DIRECTORS.

Table of Contents

PROPOSAL IV:

THE APPROVAL OF THE AMENDMENT AND RESTATEMENT OF THE EMCORE CORPORATION 2010 EQUITY INCENTIVE PLAN

On March 2, 2011, upon unanimous recommendation of the Compensation Committee, the Board of Directors of the Company adopted the amendment and restatement of the EMCORE Corporation 2010 Equity Incentive Plan (the "Amended Plan"), subject to the approval of our shareholders. The primary purpose of the Amended Plan is to: (i) increase by 3,000,000 the number of shares of Common Stock available for grant under the Amended Plan from 4,000,000 to 7,000,000 shares of Common Stock, (ii) permit the granting of performance stock units or performance units, which may qualify as "performance-based" compensation under Section 162(m) of the Internal Revenue Code of 1986, as amended (the "Code"), (iii) permit the granting of stock appreciation rights, restricted stock units, unrestricted share awards, stock purchase rights, and deferred stock units, (iv) permit the granting of restricted stock and restricted stock units with a vesting period of less than three years for an aggregate number of shares of Common Stock not to exceed 350,000, (v) provide the Compensation Committee the ability to adjust or eliminate change in control provisions for future awards , and (vi) make certain other administrative changes.

On April 7, 2010, upon recommendation of the Compensation Committee, the Board of Directors of the Company adopted the EMCORE Corporation 2010 Equity Incentive Plan (the "2010 Plan"). The 2010 Plan became effective upon its approval by the Company's shareholders at the 2010 Annual Meeting, and has been employed as a principal feature of the Company's compensation program since that time. As of April 20, 2011, [______] shares have been issued under the 2010 Plan and only [______] shares remain available for future issuance. Furthermore, no shares are currently available for grant under the EMCORE Corporation 1995 Incentive and Non-Statutory Stock Option Plan (as amended, the "1995 Plan") or the 2000 Stock Option Plan (as amended, the "2000 Plan"). The 1995 Plan had allowed the grant of a total of 2,744,118 shares of Common Stock (on a post-split basis) pursuant to stock options and stock appreciation rights. The 2000 Plan had allowed the grant of a total of 15,850,000 shares of Common Stock upon the exercise of stock options.

Our Company's philosophy on employee compensation is to provide employees and management with equity participation linked to long-term stock price performance, while at the same time remaining sensitive to the potential impact on our other shareholders. We believe that offering broad-based equity compensation will assist the Company in achieving its goals to foster and promote the long-term financial success of the Company and enhance shareholder value by (a) motivating superior performance by employees and directors, (b) providing employees and directors with an ownership interest in the Company, and (c) enabling the Company to attract and retain the services of outstanding employees and directors upon whose judgment, interest, and special effort the successful conduct of its operations is largely dependent, while also receiving a federal income tax deduction for certain compensation paid under the Amended Plan under Section 162(m) of the Code. The Board of Directors believes that the ability to offer a variety of forms of equity compensation—including grants of stock options, stock appreciation rights, deferred stock units, performance stock, performance stock units, performance units, restricted stock, restricted stock units, stock purchase rights, and unrestricted share awards— is necessary for us to attract and retain the services of well-qualified employees and advisors, including officers and directors who will contribute to our success. Equity compensation has historically formed a significant portion of our employees' overall compensation, and almost all of our current employees have received equity compensation in the form of options and/or restricted stock awards.

The Board of Directors recommends approval of the Amended Plan. If the shareholders approve the Amended Plan, it will allow our Company to offer an increased variety of equity incentives and to provide sufficient reserves of shares, based on our current business plans, to ensure the Company's ability to continue to provide new hires, employees, and management with an equity stake in the Company over the next year. The Board of Directors believes that adopting the Amended Plan is in the best interests of the Company and the shareholders. In the event this proposal is not approved by our shareholders, and as a consequence we are unable to continue to make equity grants at competitive

levels, the Board of Directors believes that it will negatively affect our ability to meet our need for highly qualified personnel and our ability to manage future growth.

A description of the material provisions of the Amended Plan is set forth below. The statements made in this Proposal IV concerning terms and provisions of the Amended Plan are summaries and do not purport to be a complete recitation of the Amended Plan provisions. These statements are qualified in their entirety by express reference to the full text of the Amended Plan, including as it is proposed to be adopted. The full text of the Amended Plan is attached to this Proxy Statement as Exhibit B and is incorporated by reference herein.

As of April 20, 2011, the closing price of our Common Stock was \$[____].

Administration and Eligibility

The Amended Plan will be administered by the Compensation Committee or, in certain instances, its designee. Employees, non-employee directors, and consultants of the Company and its subsidiaries will be eligible to receive awards of Common Stock, stock options, stock appreciation rights, performance stock, performance stock units, performance units, restricted stock, restricted stock units, stock purchase rights, or deferred stock units at the Compensation Committee's discretion. There are approximately 650 employees and six non-employee directors eligible to receive awards under the Amended Plan. Subject to applicable law, the Compensation Committee may delegate to an officer, director or group of officers or directors of the Company or its affiliates some or all of its authority under the Amended Plan with respect to participants who are not our executive officers.

Shares Available for Issuance

If the Amended Plan is approved by shareholders, 7,000,000 shares will be authorized for issuance under the Amended Plan. A participant may receive in any 12 month period a maximum award of stock options, stock appreciation rights, and/or stock purchase rights covering 500,000 shares of Common Stock and a maximum award of performance stock, performance stock units, restricted stock, restricted stock units, deferred stock units, and/or share awards covering 200,000 shares of Common Stock, A participant may receive in any calendar year a certain maximum amount of cash earned in connection with the grant of performance units. The maximum number of shares that may be issued with respect to incentive stock options is 7,000,000. Awards that for any reason are canceled, terminated, forfeited, settled in cash or otherwise settled without the issuance of Common Stock will be available again under the Amended Plan. In the event of a stock dividend, stock split, share combination, extraordinary cash dividend, recapitalization, reorganization, merger, consolidation, split-up, spin-off, combination, exchange of shares, or other similar event affecting the Company's Common Stock, the Compensation Committee will adjust the shares available under the Amended Plan and any outstanding awards to reflect the event and preserve the intrinsic value of the awards. The Compensation Committee does not have the power to reduce the exercise price of outstanding options or base price of any outstanding stock appreciation rights or grant any new award or cash payment in substitution or upon cancellation of options or stock appreciation rights for any other reason unless the adjustment is approved by our shareholders.

Amendment or Termination

The Board or Compensation Committee may terminate, amend or suspend the Amended Plan at any time. If the proposal is approved by our shareholder, the Amended Plan will continue in effect until June 14, 2021 (or if applicable, the 10 year anniversary date of the latest shareholder approval of the Amended Plan), if not earlier terminated by the Board or Compensation Committee. An amendment to the Amended Plan will be submitted for shareholder approval to the extent required by the Code or other applicable laws, rules or regulations or if the amendment will (i) materially increase the benefits under the Amended Plan, (ii) materially increase the number of shares of Common Stock subject to the Amended Plan or the individual award limitations set forth in the Amended Plan, other than for antidilutive purposes, (iii) modify the restrictions on repricing set forth on the Amended Plan or (iv) materially modify the requirements for participation in the Amended Plan.

Stock Options, Stock Appreciation Rights and Stock Purchase Rights

Stock options granted under the Amended Plan may be incentive stock options (within the meaning of Section 422 of the Code) or non-statutory stock options. The grant date of options granted under the Amended Plan will be the date the options are awarded by the Compensation Committee except with respect to certain options granted to new hires

in accordance with Company policy, which are granted on the date the new employee begins work at the Company. Stock options will have an exercise price per share that is no less than the fair market value (as defined in the Amended Plan) of a share of Common Stock as reported by Nasdaq on the date that the options are granted.

Table of Contents

Options under the Amended Plan will vest based on a minimum period of service or the occurrence of certain events, as determined by the Compensation Committee. No option will remain exercisable beyond 10 years after its grant date.

Stock appreciation rights may be granted to participants in tandem with options or on their own. Unless otherwise determined by the Compensation Committee at or after the grant date, tandem stock appreciation rights will have substantially similar terms as the options with which they are granted. The grant date of stock appreciation rights granted under the Amended Plan will be the date the stock appreciation rights are awarded by the Compensation Committee or a future date determined by the Compensation Committee. Free-standing stock appreciation rights will vest based on a minimum period of service or the occurrence of certain, specified events, as determined by the Compensation Committee. No stock appreciation right will remain exercisable longer than 10 years after its grant date.

Unless otherwise determined by the Compensation Committee at or after the grant date, in the event of a participant's death or disability, the participant's unvested options or stock appreciation rights will vest and all of the participant's options and stock appreciation rights will remain exercisable until the second anniversary of the participant's termination of employment (or the expiration of the award's term, whichever is earlier). If a participant's employment is terminated for cause, all of the participant's options and stock appreciation rights will immediately be forfeited and canceled. If a participant's employment is terminated for any reason other than death, disability or for cause, then the participant may exercise any options or stock appreciation rights that are exercisable on the date of such termination until the earlier of (i) the 90th day following the date of such termination or, if later, the 90th day following expiration of any blackout period in effect with respect to such options or stock appreciation rights, and (ii) the expiration of the term of such options or stock appreciation rights, and any options or stock appreciation rights that are not exercisable upon the participant's termination or retirement shall be forfeited and canceled as of the date of such termination.

Stock Purchase Rights may be granted to participants in tandem with options or on their own. Stock Purchase Rights may be granted to participants at such time or times as shall be determined by the Compensation Committee. Each award of Stock Purchase Rights shall specify the terms, conditions and restrictions related to the offer, including the number of shares of Common Stock that the participant shall be entitled to purchase, the price to be paid, and the time within which the participant must accept such offer. Unless the Compensation Committee determines otherwise, the Company shall have a repurchase option exercisable upon the voluntary or involuntary termination of the participant's service with the Company for any reason. The purchase price for shares of Common Stock repurchased shall be the original price paid by the participant. The repurchase option shall lapse at a rate determined by the Compensation Committee.

Performance Stock, Performance Stock Units, Performance Units

Performance stock is Common Stock of the Company that is subject to forfeiture until predetermined performance conditions have been achieved. A performance stock unit is a contractual right to receive a stated number of shares of Common Stock, or if provided by the Compensation Committee on or after the grant date, cash equal to the fair market value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate fair market value equal to such stated number of shares of Common Stock, which right is forfeitable until the achievement of predetermined performance conditions. A performance unit is a contractual right to receive a cash-denominated award, payable in cash or shares of Common Stock or a combination thereof, which right is forfeitable until the achievement of predetermined performance conditions.

The grant date of any performance stock, performance stock units, or performance units granted under the Amended Plan will be the date on which such performance stock, performance stock units, or performance units are awarded by the Compensation Committee or on such other future date as the Compensation Committee shall determine.

Performance stock, performance stock units, and performance units granted under the Amended Plan will vest based on the achievement of pre-determined performance goals over performance periods determined by the Compensation Committee or upon the occurrence of certain events, as determined by the Compensation Committee. Unless otherwise determined by the Compensation Committee at or after the grant date, in the event of a participant's death or disability, a pro rata portion of the participant's performance stock, performance stock units, and performance units will vest to the extent performance goals are achieved at the end of the performance period.

Table of Contents

Unless otherwise determined by the Compensation Committee at or after the grant date, upon a termination of employment for any other reason, all outstanding performance stock, performance stock units, and performance units held by the participant will be immediately canceled.

Restricted Stock, Restricted Stock Units, Share Awards

Restricted stock is Common Stock of the Company that is subject to forfeiture until vested. A restricted stock unit is a contractual right to receive a stated number of shares of Common Stock, or if provided by the Compensation Committee on or after the grant date, cash equal to the fair market value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate fair market value equal to such stated number of shares of Common Stock, that is subject to forfeiture until vested. Share awards are awards of unrestricted Common Stock.

The grant date of any restricted stock or restricted stock unit under the Amended Plan will be the date on which such restricted stock or restricted stock units are awarded by the Compensation Committee or on such other future date as the Compensation Committee shall determine. Restricted stock and restricted stock units granted under the Amended Plan will vest based on a minimum period of service (generally over a three-year period) or the occurrence of events specified by the Compensation Committee. The Compensation Committee has the discretion to grant restricted stock and restricted stock unit awards that have a vesting period of less than three years for an aggregate number of shares of Common Stock not to exceed 350,000 shares (subject to adjustment if awards of such stock or units are forfeited). Unless otherwise determined by the Compensation Committee at or after the grant date, in the event of a participant's death or disability, a pro rata portion of the participant's restricted stock or restricted stock units will vest, and the remainder will be forfeited. Unless otherwise determined by the Compensation Committee at or after the grant date, upon a termination of employment for any other reason, any unvested restricted stock or restricted stock units of the participant will be canceled. Share awards are granted by the Compensation Committee upon terms and conditions determined by the Compensation Committee in its discretion.

Deferred Stock Units

Each deferred stock unit granted under the Amended Plan represents the contractual right to receive a stated number of shares of Common Stock or, if provided by the Compensation Committee on or after the grant date, cash equal to the fair market value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate fair market value equal to such stated number of shares of Common Stock, on a specified future date. The grant date of any freestanding deferred stock units under the Amended Plan will be the date on which such freestanding deferred stock units are awarded by the Compensation Committee or on such other future date as the Compensation Committee shall determine. Deferred stock units may be granted by the Compensation Committee independent of other awards or compensation, or, to the extent permitted by law and subject to the terms and conditions determined by the Compensation Committee, they may be obtained at the participant's election instead of cash compensation. Generally, upon a participant's termination of employment other than for cause, the Company will issue to the participant the shares of Common Stock underlying any of the participant's deferred stock units. If a participant's employment terminates for cause, any deferred stock units granted independently by the Compensation Committee will be immediately canceled.

Change in Control

Upon the occurrence of a change in control of the Company (as defined in the Amended Plan), unless outstanding awards are honored, assumed or substituted with alternative awards that provide substantially similar terms, conditions and economic value to the substituted awards, all awards will immediately become exercisable and any restrictions related to the awards will lapse, provided, that, at the discretion of the Compensation Committee (as

constituted immediately prior to the change in control) each option, stock appreciation right, restricted stock unit, and/or deferred stock unit may be canceled in exchange for an amount of cash calculated pursuant to the Amended Plan. Notwithstanding the foregoing, performance stock, performance stock units, and performance units that are outstanding upon the occurrence of such a change in control will be treated as provided in the individual award agreement governing such award, and the Compensation Committee may provide for different terms and provisions in the individual award agreements with respect to other awards.

New Plan Benefits and Option History

Future benefits under the Amended Plan are not currently determinable. Moreover, it is also not possible to determine the amounts that would have been paid for fiscal 2010 had the Amended Plan been in effect during such year.

Federal Income Tax Consequences

Options

The grant of an option under the Amended Plan generally will not give rise to any tax consequences for the participant or the Company.

Incentive Stock Options

A participant generally will have no taxable income upon exercise of an incentive stock option, provided the employee remains an employee of EMCORE at all times from the grant date of the option to the day three months before such exercise, except that the alternative minimum tax may apply. Subject to certain statutory restrictions, gain realized upon a disposition of the Company's Common Stock received pursuant to the exercise of an incentive stock option will generally be taxed as long-term capital gain if the participant holds the shares for at least two years after the date of grant and one year after the date of exercise (the "holding period requirement"). Generally, if the participant has not satisfied the holding period requirement, the participant will recognize ordinary income upon the disposition of the Common Stock equal to the excess of the fair market value of the Common Stock at the time the option was exercised over the exercise price (but not in excess of the gain realized on the sale). The balance of the realized gain, if any, generally will be capital gain. An employee who dies while holding an incentive stock option is entitled to beneficial tax treatment upon exercise and disposition of the option regardless of whether the employee satisfies the employment or holding period requirements. The Company will not be entitled to a deduction with respect to the exercise of an incentive stock option, although the Company generally will be entitled to a deduction to the extent the participant recognizes ordinary income.

Nonqualified Stock Options

Upon exercise of a non-qualified stock option, a participant generally will recognize ordinary income equal to the difference between the fair market value of the shares acquired and the exercise price. Upon a disposition of the shares acquired by the exercise of a non-qualified option, the participant generally will have taxable capital gain or loss, measured by the difference between the amount realized on the disposition and the tax basis of the shares of Common Stock (generally, the amount paid for the shares plus the amount treated as ordinary income at the time the option was exercised). The Company generally will be entitled to a tax deduction equal to the amount recognized as ordinary income by the participant in connection with an option, stock appreciation rights, or other award, but will be entitled to no tax deduction relating to amounts that represent a capital gain to a participant.

Section 162(m)

Section 162(m) of the Code places a limit of \$1,000,000 per person on the amount the Company may deduct in any one year for compensation paid to its CEO and the next three highest compensated officers (other than the Chief Financial Officer); provided, however, that Section 162(m) of the Code generally allows a company to obtain tax deductions without limit for performance-based compensation. The Company intends that options and stock appreciation rights, and, subject to shareholder approval of the performance goals applicable to performance-based awards, performance stock, performance stock units, and performance units granted under the Amended Plan will qualify as performance-based compensation not subject to Code Section 162(m)'s \$1 million deductibility cap. A

number of requirements must be met in order for particular compensation to so qualify, however, so there can be no assurance that such compensation will be fully deductible in all circumstances. In addition, other awards under the Amended Plan, such as restricted stock and restricted stock units, generally do not qualify for the performance-based compensation exception, so that compensation paid to executive officers in connection with such awards may not be deductible.

Performance Objectives that May be Applied Under the Amended Plan

As described above, certain awards under the Amended Plan may be subject to performance objectives. Performance objectives applicable to awards intended to qualify as performance-based compensation under Section 162(m) of the Code will be based on the relative or comparative achievement of one or more of the following criteria, whether in absolute terms or relative to the performance of one or more similarly situated companies or a published index covering the performance of a number of companies: net sales; revenue; revenue growth or product revenue growth; operating income (before or after taxes); pre- or after-tax income (before or after allocation of corporate overhead and bonus); net earnings; earnings per share; net income (before or after taxes); return on equity; total shareholder return; return on assets or net assets; appreciation in and/or maintenance of share price; market share; gross profits; earnings (including adjusted pre-tax earnings, earnings before taxes, earnings before interest and taxes or earnings before interest, taxes, depreciation and amortization); economic value-added models or equivalent metrics; comparisons with various stock market indices; reductions in costs; total net cash flow; cash flow or cash flow per share (before or after dividends); return on capital (including return on total capital or return on invested capital); cash flow return on investment; improvement in or attainment of expense levels or working capital levels; operating margins, gross margins or cash margin; year-end cash; debt reductions; shareholder equity; market share; regulatory achievements; and implementation, completion or attainment of measurable objectives with respect to customer satisfaction, research, development, products or projects and recruiting and maintaining personnel. Performance objectives under the Amended Plan may be established on a company-wide basis or with respect to one or more business units or divisions, or subsidiaries.

The foregoing objectives may include or exclude any or all "extraordinary items" as determined under generally accepted accounting principles in the United States of America and as identified in the financial statements, notes to the financial statements or management's discussion and analysis in the most recent report filed with the Securities and Exchange Commission, including, without limitation, the charges or costs associated with restructurings of the Company, discontinued operations, other unusual or non recurring items, and the cumulative effects of accounting changes. Under the Amended Plan, the Compensation Committee is not permitted to exercise its discretion with respect to performance objectives for awards to covered executives intended to be "performance-based compensation" under Section 162(m) of the Code if doing so (or if the ability to do so) would cause the award to fail to qualify as "performance-based" compensation under Section 162(m) of the Code.

The foregoing general tax discussion is intended for the information of shareholders considering how to vote with respect to this proposal and not as tax guidance to participants in the Amended Plan. Participants in the Amended Plan are strongly urged to consult their own tax advisors regarding the federal, state, local, foreign and other tax consequences to them of participating in the Amended Plan.

RECOMMENDATION OF THE BOARD OF DIRECTORS

THE BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS THAT SHAREHOLDERS VOTE "FOR" THE APPROVAL OF THE AMENDMENT AND RESTATEMENT OF THE EMCORE CORPORATION 2010 EQUITY INCENTIVE PLAN.

PROPOSAL V:

INCREASE IN THE NUMBER OF SHARES RESERVED FOR ISSUANCE UNDER THE EMCORE 2000 EMPLOYEE STOCK PURCHASE PLAN

The maximum number of shares of our Common Stock that may be issued pursuant to purchases made under the 2000
Employee Stock Purchase Plan (the "2000 ESPP") is currently 4,500,000 shares. As of April 20, 2011, [
shares have been issued under the 2000 ESPP and [] shares remain available for future issuance. We use
the 2000 ESPP to facilitate the development of equity ownership by our employees, which we believe more
effectively aligns the interests of our employees with those of our shareholders.

On November 8, 1999, the Board of Directors adopted the EMCORE Corporation 2000 Employee Stock Purchase Plan, which provides the Company's employees with the opportunity to acquire an ownership interest in EMCORE Corporation through the purchase of shares of the Company's Common Stock through payroll deductions. The option price is set at 85% of the market price for the Company's Common Stock on either the first or last day of the participation period, whichever is lower. Contributions are limited to 10% of an employee's compensation. The 2000 ESPP became effective upon its approval by the Company's shareholders at the 2000 Annual Meeting. It was amended by a vote of the shareholders at the Company's 2009 Annual Meeting to increase the number of shares available for issuance from 2,000,000 to 4,500,000. It was previously amended by a vote of the shareholders at the Company's 2006 Annual Meeting to increase the number of shares available for issuance from 1,000,000 to 2,000,000. In fiscal 2004, the 2000 ESPP was amended by the Board of Directors to change from a 12-month duration plan to a 6-month duration plan, with new participation periods beginning in January and July of each year. In fiscal 2011, the 2000 ESPP was amended by the Board of Directors to change the offering periods so that they begin in February and August of each year, as further described below.

The 2000 ESPP currently provides for a total of 4,500,000 shares of the Company's Common Stock for purchase by employees, subject to adjustment for certain changes in our capital (described under "Changes in Capital" below). The 2000 ESPP qualifies as an "employee stock purchase plan" under section 423 of the Internal Revenue Code of 1986, as amended (the "Internal Revenue Code"), so that our employees may enjoy certain tax advantages (see "Certain Federal Income Tax Consequences" below).

On March 2, 2011, the Board of Directors, acting on the recommendation of the Compensation Committee, unanimously adopted an amendment to the 2000 ESPP, subject to approval by the shareholders, to increase the total number of shares of Common Stock on which options may be granted under the 2000 ESPP by 2,500,000, to 7,000,000. The Board of Directors recommends approval of this amendment to the 2000 ESPP to permit the issuance of this increased number of shares of Common Stock thereunder. The Board of Directors believes that this proposed increase is in the best interests of the Company and the shareholders. In the event this proposal is not approved by our shareholders, and, as a consequence, we are unable to continue to grant options at competitive levels, the Board of Directors believes that our ability to encourage stock ownership in the Company, thus aligning the interests of the employees with those of the shareholders, will be materially impaired.

If this proposal is adopted, the first sentence of Section 5.01(a) of the 2000 ESPP would be amended to read, in its entirety, as follows:

"The maximum number of shares of Common Stock that may be issued under the Plan shall be 7,000,000 shares."

This proposal summarizes the essential features of the 2000 ESPP, as it would be amended pursuant to this proposal. You should read the amended plan for a full statement of its terms and conditions. The full text of the 2000 ESPP is attached to this Proxy Statement as Exhibit C and is incorporated by reference herein.

Description of Material Features of the 2000 ESPP

Administration. The Board of Directors selects at least three of its members to serve on a Committee that administers the 2000 ESPP. Subject to limitations of applicable laws or rules, the Board of Directors may exercise the powers of the Committee, and, if no such committee exists, the Board of Directors will perform all the functions of the Committee. All decisions and actions of the Committee will be final and conclusive. Subject to limitations of applicable laws or rules, the Committee may delegate its administrative responsibilities and powers under the 2000 ESPP.

Table of Contents

In addition to its other powers under the 2000 ESPP described in this summary, and subject to the express provisions of the 2000 ESPP, the Committee will have discretionary authority to:

- interpret the 2000 ESPP and option agreements,
- determine eligibility to participate in the 2000 ESPP,
- adjudicate and determine all disputes arising under or in connection with the 2000 ESPP,
- impose restrictions on ownership and transferability of the shares of our Common Stock underlying options granted under the 2000 ESPP.
 - establish procedures for carrying out the 2000 ESPP, and
 - make all other determinations deemed necessary or advisable for administering the 2000 ESPP.

The Compensation Committee (which consists of Messrs. Bogomolny, Gillen, McCorkle and Scott, each of whom is a director, but not an employee, of EMCORE) has administrative authority over the 2000 ESPP.

Eligibility. All full-time and part-time employees of EMCORE and those of our designated subsidiaries are eligible to participate in the 2000 ESPP, except:

an employee may not be granted an option under the 2000 ESPP if:

- immediately after the grant of such option, the employee would own 5 percent or more of the vote or value of all classes of our stock or the stock of any of our subsidiaries, or
- such option would permit the employee to purchase more than \$25,000 of our stock (using the fair market value of our stock at the time the option is granted) under the 2000 ESPP (and any other employee stock purchase plan of us or our subsidiaries) per calendar year when the option is outstanding;

and the Committee may, in its discretion, exclude from participation in the 2000 ESPP employees who:

- customarily work 20 or fewer hours per week, customarily work 5 or fewer months per calendar year, or
- are highly compensated employees (within the meaning of Section 414(q) of the Internal Revenue Code).

Since the effective date of the 2000 ESPP (and until the Committee determines otherwise), employees who customarily work 20 or fewer hours per week, or who customarily work 5 or fewer months per calendar year, have been ineligible to participate in the 2000 ESPP.

Approximately 650 employees are currently eligible to participate in the 2000 ESPP.

Terms of Options.

Options and Offering Periods. An option granted to an eligible employee under the 2000 ESPP allows the employee to use payroll deductions accumulated during successive six-month offering periods to purchase shares of our Common Stock at the end of each offering period. The option price of the shares is the lesser of 85 percent of our Common Stock's fair market value on the first day of the offering period or the last day of the offering period.

begin on the first trading date on or after February 25 and August 26, and end on the last trading date on or before August 25 and February 24 of each calendar year, respectively, while the 2000 ESPP is in effect. The Committee may change the commencement and duration of offering periods under the 2000 ESPP. Our Board of Directors also may terminate a pending offering period, in which case payroll deductions that have accumulated in participants' accounts (see "Payroll Deductions" below) will be used to exercise outstanding options or returned to the appropriate participants, as determined by the Board of Directors, in its discretion.

Participation. Each eligible employee decides for himself or herself whether to participate or not participate in the 2000 ESPP during each offering period. An eligible employee may elect to enroll in the 2000 ESPP by filing an agreement with the Company's payroll office before the first day of the applicable offering period.

Payroll Deductions. A participant's agreement must specify the percentage, from 1 to 10 percent, to be deducted from his or her compensation (as defined in the 2000 ESPP) on each payroll date during the offering period. Payroll deductions will be credited to a bookkeeping account in the participant's name. The Company does not set aside any assets with respect to such participant accounts, and such accounts do not bear interest. A participant may decrease his or her contribution rate no more than once each offering period. The Committee may limit the number of participants who change their contribution rates during any offering period and may, subject to certain limitations in the 2000 ESPP, decrease the contribution rate of any participants. Except in the event of a change in control of EMCORE (as described under "Changes in Capital" below), participants are not permitted to make contributions to their accounts under the 2000 ESPP otherwise than through payroll deductions as described above.

Exercise of Option. Unless a participant provides the Company with written notice or withdraws from the 2000 ESPP, his or her option will be automatically exercised on the last day of the offering period to purchase the maximum number of full shares of our Common Stock that can be purchased at the applicable option price using the accumulated payroll deductions in the participant's account. The 2000 ESPP sets forth certain limitations on the number of shares that a participant may purchase in a single offering period. Any excess payroll deductions remaining in a participant's account after exercise of his or her option will be returned to the participant, without interest, and may not be used to exercise options granted under the 2000 ESPP in any subsequent offering period (except for any excess funds attributable to the inability to purchase a fractional share, which will be retained in the participant's account for a subsequent offering period or may be withdrawn by the participant).

Withdrawal/Termination of Employment. A participant may withdraw from the 2000 ESPP at any time, receiving payment of his or her accumulated payroll deductions and ceasing further payroll deductions, by providing the Company with written notice to withdraw. If a participant terminates his or her employment, such participant will be considered to have withdrawn from the 2000 ESPP. A leave of absence in excess of 90 days without a guaranteed right to reemployment will be considered a termination of employment for purposes of the 2000 ESPP. When a participant withdraws from the 2000 ESPP, his or her unexercised options will automatically terminate, and we will return to the participant all accumulated payroll deductions in his or her account.

Transferability of Options. No one other than the participant who receives an option under the 2000 ESPP may exercise such option during such participant's lifetime. Participants are not entitled to transfer, assign or otherwise dispose of their payroll deductions or rights to exercise options or receive Common Stock under the 2000 ESPP, except, in the event of a participant's death, by will, the laws of descent and distribution or to the deceased participant's designated beneficiary.

Changes in Capital. In the event of certain changes in our outstanding Common Stock or capital structure, such as a stock dividend, stock split, recapitalization, reorganization, merger, consolidation, or corporate separation or division, or change in the number of shares of our capital stock effected without receipt of full consideration, the Committee may, in its discretion, make appropriate adjustments or substitutions with respect to the following to reflect equitably the effects of such changes to participants in the 2000 ESPP:

- the number, class and kind of shares available under the 2000 ESPP,
- the number, class and kind of shares covered by outstanding options,
- the maximum number of shares that a participant may purchase during an offering period,

- the option prices of outstanding options, and
- any other necessary characteristics or terms of the 2000 ESPP or the options.

If a "change in control" of EMCORE (as defined in the 2000 ESPP) occurs, the 2000 ESPP gives the Committee discretion to:

- •terminate the pending offering period and permit each participant to make a one-time cash contribution equal to the amount that the Committee determines such participant would have contributed under the 2000 ESPP through payroll deductions until the otherwise scheduled end of the pending offering period and use the accumulated payroll deductions to exercise outstanding options; or
- •terminate each participant's options in exchange for a cash payment equal to (a) the balance of the participant's account under the 2000 ESPP, plus (b) the highest value of the consideration received for a share of our Common Stock in the change in control transaction (or, if greater, the highest fair market value of a share of our Common Stock during the 30 consecutive trading days prior to the closing or expiration date of the change in control transaction), less the option price of the participant's option (determined as if the option were exercised on the closing or expiration date of the change in control transaction), multiplied by the number of full shares of our Common Stock that the participant could have purchased immediately prior to the change in control with the then outstanding balance of the participant's account under the 2000 ESPP.

Tax Withholding Obligations. If any taxes are required to be withheld when a participant exercises his or her option, or when shares are issued under the 2000 ESPP or disposed of by a participant, we may, as a condition to delivery of stock certificates under the 2000 ESPP, require that the participant remit to us the amount necessary to satisfy such taxes, or we may make other arrangements, including withholding from the participant's compensation or other amounts due to such participant, to satisfy such taxes.

Amendment and Termination of the 2000 ESPP. Our Board of Directors may terminate, discontinue, amend or suspend the 2000 ESPP at any time. However, without approval of the shareholders, the Board of Directors may not:

- increase the maximum number of shares that we may issue under the 2000 ESPP, or that a participant may purchase in any offering period (except as described under "Changes in Capital" above);
- •change the class of employees eligible to receive options under the 2000 ESPP (except for the designation of any subsidiaries whose employees will be eligible to participate in the 2000 ESPP); or
 - change the formula by which the option price is determined under the 2000 ESPP.

Except for an amendment or termination described under "Changes in Capital" above, or in the last sentence of the portion of this summary under "Terms of Options - Options and Offering Periods," above, no amendment or termination of the 2000 ESPP may materially adversely affect the existing rights of any participant under his or her option without such participant's consent.

Certain Federal Income Tax Consequences. The following is a brief summary of certain significant United States Federal income tax consequences under the Internal Revenue Code, as in effect on the date of this summary, applicable to EMCORE and its employees in connection with participation and purchase of shares of our Common Stock under the 2000 ESPP. This summary is not intended to be exhaustive, and among other things, does not describe state, local or foreign tax consequences, or the effect of gift, estate or inheritance taxes. References to "EMCORE" and "us" in this summary of tax consequences mean EMCORE Corporation or any subsidiary of EMCORE Corporation that employs an employee who participates in the 2000 ESPP, as the case may be.

An employee will not recognize any taxable income upon an election to participate in the 2000 ESPP and receipt of an option to purchase stock under the 2000 ESPP. The amounts deducted from the salary of an employee who

participates in the 2000 ESPP will constitute ordinary income taxable to the employee. The 2000 ESPP is intended to qualify for the favorable income tax consequences of Section 423 of the Internal Revenue Code. As such, no income tax consequences will arise for an employee when shares of our Common Stock are purchased by exercising such employee's option under the 2000 ESPP, provided the employee remains an employee of EMCORE at all times from the grant date of the option (i.e., the first day of the offering period) to the day three months before such exercise. The employee receives a tax basis in the shares purchased equal to his or her payroll deductions used to exercise the option.

The tax consequences arising from a disposition of shares acquired from the exercise of an option granted under the 2000 ESPP will depend on whether the employee meets certain holding period requirements. Generally, although all dispositions of shares acquired pursuant to the 2000 ESPP may generate both ordinary income and capital gains, the ordinary income arising from a "qualifying disposition" is lower than it would be if such disposition were not a "qualifying disposition." In order to satisfy the requirements for a "qualifying disposition," a participating employee, among other things, must hold his or her shares until at least the latter of (i) two years after the grant date of the employee's option (which for income tax purposes is deemed to be the first day of the offering period) and (ii) one year after the date of such purchase. A disposition is treated as a "qualifying disposition" if the employee died while owning such shares.

If the requirements for a "qualifying disposition" are met, the ordinary income component of the employee's income from the disposition of such shares (or upon the employee's death) will generally be limited to the lesser of (a) the excess of the fair market value of the shares when the option was granted (i.e., the first day of the offering period) over the purchase price paid for such shares, or (b) the excess of the fair market value of such shares at the time of such disposition (or death) over the purchase price paid for the shares. Generally speaking, this rule is intended to treat, as ordinary income, the proceeds (if any) received by an employee from the sale of shares acquired pursuant to the 2000 ESPP to the extent that the exercise price of the option to acquire the shares is less than (but at least 85% of) the fair market value of the shares when the option was granted. Such amount will be treated as compensation and taxed at ordinary income tax rates, and will increase the employee's basis in such shares. Any proceeds received in excess of the employee's adjusted basis in the shares would be taxable to the employee at capital gains tax rates. If an employee sells such shares for less than the purchase price paid, he or she will recognize no such ordinary income, and such employee will have a capital loss equal to the difference between the sale price and the purchase price previously paid. EMCORE is not entitled to a tax deduction with respect to any such disposition.

If the requirements for a "qualifying disposition" are not met, the employee will be required to report taxable ordinary income at the time of such disposition to the extent of the difference between the fair market value of such shares on the date of the exercise of the option (rather than, in the case of a "qualifying disposition," the fair market value of the shares on the date of the grant of the option) and the purchase price paid for the shares. EMCORE will generally be allowed a tax deduction equal to the amount of such ordinary income so reported by such employee. The basis of an employee in such shares acquired under the 2000 ESPP will be increased by such amount reported as ordinary income by such employee upon disposition of such shares. Any proceeds received for the shares in excess of such employee's adjusted basis will be taxable as capital gain; if such adjusted basis exceeds the amount received for such shares, such excess will be a capital loss.

RECOMMENDATION OF THE BOARD OF DIRECTORS

THE BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS THAT SHAREHOLDERS VOTE "FOR" THE INCREASE IN SHARES AVAILABLE UNDER THE 2000 EMPLOYEE STOCK PURCHASE PLAN FROM 4,500,000 TO 7,000,000 IN ACCORDANCE WITH PROPOSAL V.

PROPOSAL VI: ADVISORY VOTE ON EXECUTIVE COMPENSATION

In accordance with the Dodd-Frank Wall Street Reform and Consumer Protection Act, enacted in July 2010, and the related compensation disclosure rules of the Securities and Exchange Commission we are asking our shareholders to vote to approve, on a non-binding, advisory basis, the compensation of our Named Executive Officers as disclosed in this Proxy Statement.

As described in detail under the heading "Compensation Discussion and Analysis," we seek to closely align the interests of our Named Executive Officers with the interests of our shareholders. Our compensation programs are designed to reward our Named Executive Officers for the achievement of short-term and long-term strategic and operational goals, while at the same time avoiding the encouragement of unnecessary or excessive risk-taking.

The vote on this resolution is not intended to address any specific element of compensation; rather, the vote relates to the compensation of our Named Executive Officers, as described in this Proxy Statement. The vote is advisory, which means that the vote is not binding on the Company, our Board of Directors or the Compensation Committee of the Board of Directors. Although the vote is non-binding, the Compensation Committee will consider the voting results when it evaluates whether any actions are necessary regarding our compensation program.

Accordingly, we ask our shareholders to approve the following resolution at the Annual Meeting:

"RESOLVED, that the Company's shareholders approve, on an advisory basis, the compensation awarded to the Corporation's Named Executive Officers as disclosed in the CD&A, the accompanying compensation tables, and the related narrative disclosure in this Proxy Statement."

RECOMMENDATION OF THE BOARD OF DIRECTORS

THE BOARD OF DIRECTORS UNANIMOUSLY RECOMMENDS A VOTE FOR THE APPROVAL OF THE COMPENSATION OF OUR NAMED EXECUTIVE OFFICERS, AS DISCLOSED IN THIS PROXY STATEMENT.

PROPOSAL VII:

ADVISORY VOTE ON THE FREQUENCY OF AN ADVISORY VOTE ON EXECUTIVE COMPENSATION

In accordance with the Dodd-Frank Wall Street Reform and Consumer Protection Act and the related rules of the Securities and Exchange Commission, we are asking our shareholders to vote, on a non-binding, advisory basis, for their preference as to how frequently we should seek future advisory votes on the compensation of our Named Executive Officers. By voting with respect to this Proposal VII, shareholders may indicate whether they would prefer that we conduct future advisory votes on executive compensation once every one, two, or three years. Shareholders also may, if they wish, abstain from casting a vote on this proposal.

Our Board of Directors has determined that an annual advisory vote on executive compensation will allow our shareholders to provide timely, direct input on the Company's executive compensation philosophy, policies and practices as disclosed in the proxy statement each year. The Board believes that an annual vote is therefore consistent with the Company's efforts to engage in an ongoing dialogue with our shareholders on executive compensation and corporate governance matters.

The Company recognizes that the shareholders may have different views as to the best approach for the Company, and therefore we look forward to hearing from our shareholders as to their preferences on the frequency of an advisory vote on executive compensation.

This vote is advisory and not binding on the Company or our Board of Directors. However, the Board of Directors and the Compensation Committee will take into account the outcome of the vote when considering the frequency of future advisory votes on executive compensation. The Board may decide that it is in the best interests of our shareholders and the Company to hold an advisory vote on executive compensation more or less frequently than the frequency receiving the most votes cast by our shareholders.

The proxy card provides shareholders with the opportunity to choose among four options (holding the vote every one, two or three years, or abstaining) and, therefore, shareholders will not be vot–ing to approve or disapprove the recommendation of the Board of Directors.

RECOMMENDATION OF THE BOARD OF DIRECTORS

THE BOARD OF DIRECTORS RECOMMENDS THAT YOU VOTE FOR THE OPTION OF ONCE EVERY YEAR AS THE PREFERRED FREQUENCY FOR ADVISORY VOTES ON EXECUTIVE COMPENSATION.

GENERAL MATTERS

Annual Report on Form 10-K and Financial Statements

A shareholder may send a written request for a copy of the Company's 2010 Annual Report on Form 10-K and any additional exhibits to the Form 10-K not included in the Company's 2010 Annual Report. All such requests should be directed to the Company at 10420 Research Road, SE, Albuquerque, New Mexico 87123, Attention: Investor Relations. Following receipt of any such request by a shareholder, the Company will furnish the requested materials to the shareholder without charge. The Company's 2010 Annual Report on Form 10-K (including amendments and exhibits thereto) and this Proxy Statement are also available in the Investor Relations section of the Company's website (www.emcore.com).

Shareholder Proposals

Shareholder proposals intended to be included in our proxy materials to be distributed in connection with the 2012 Annual Meeting of Shareholders, including nominations for the Company's Board of Directors, must be received by the Company no later than January 6, 2012. Proposals should be mailed to the Company, to the attention of Alfredo Gomez, Secretary, 10420 Research Road, SE, Albuquerque, New Mexico 87123. Proposals must comply with all applicable SEC rules.

Shareholder proposals intended to be presented at the 2012 Annual Meeting of Shareholders that are not to be included in our proxy materials must comply with the requirements of our By-Laws and must be received by the Company at the address in the preceding paragraph no later than March 21, 2012.

Delivery of Documents to Shareholders Sharing an Address

The Company may deliver only one copy of the Notice of Internet Availability or, as applicable, Annual Report and Proxy Statement to shareholders who share a single address unless we have received contrary instructions from any shareholder at the address. In that case, we will deliver promptly a separate copy of the Notice of Internet Availability or, as applicable, Annual Report and/or Proxy Statement. For future deliveries, shareholders who share a single address can request a separate copy of the Company's notice of Internet availability of proxy materials or, as applicable, annual report and/or proxy statement. Similarly, if multiple copies of the notice of Internet availability of proxy materials or annual report and proxy statement are being delivered to a single address, shareholders can request a single copy of the notice of Internet availability of proxy materials or annual report and proxy statement, as applicable, for future deliveries. To make a request, please call or write to Alfredo Gomez, Secretary, EMCORE Corporation, 10420 Research Road, SE, Albuquerque, New Mexico 87123 or (505) 332-5000.

Other Matters

The Board of Directors knows of no other business which will be presented at the annual meeting. If, however, other matters are properly presented, the persons named in the enclosed proxy will vote the shares represented thereby in accordance with their judgment on such matters.

By Order of the Board of Directors,

/s/ Alfredo Gomez Alfredo Gomez, Esq. Secretary

EXHIBIT A

Certificate of Amendment of Restated Certificate of Incorporation of EMCORE Corporation

Pursuant to the provisions of Section 14A:9-2 (4) and Section 14A:9-4 (3), Corporations, General, of the New Jersey Statutes, the undersigned corporation executes the following Certificate of Amendment to its Restated Certificate of Incorporation:

- 1. The name of the corporation is: "EMCORE Corporation"
- 2. The following amendment to the Certificate of Incorporation was approved by the directors and thereafter duly adopted by the shareholders of the corporation on the 14th day of June, 2011.

Resolved, that Restated Article of the Certificate of Incorporation be amended to read as follows:

"FOURTH: The total number of shares of Capital Stock of the Corporation shall be [range between 25,882,352 and 105,882,352] shares of which:

- A. Of the Capital Stock, [range between 20,000,000 and 100,000,000] shares shall consist of Common Stock which shall be entitled to one vote per share on all matters on which holders of the Common Stock shall be entitled to vote.
- B.Of the Capital Stock, 5,882,352 shares shall consist of Preferred Stock which may be divided into such classes and such series as shall be established from time to time by resolutions of the Board of Directors and filed as an amendment to this Restated Certificate of Incorporation, without any requirement of vote or class vote of shareholders. The Board of Directors shall have the right and power to establish and designate in any such Class or Series Resolution such priorities, powers, preferences and relative, participating, optional or other special rights and qualifications, limitations and restrictions as it shall determine."

3. The number of shares outstanding at the time of the adoption of the amendment was: []. The total number of shares entitled to vote thereon was: [] shares of Common Stock
4. The number of shares voting for and against such amendment is as follows:
Number of Shares Voting for Amendment: [].
Number of Shares Voting Against Amendment : [].

5. Upon this Certificate of Amendment of the Restated Certificate of Incorporation of the Corporation becoming effective pursuant to the New Jersey Business Corporation Act (the "Effective Time"), each share of common stock of the Corporation, no par value per share (the "Old Common Stock"), issued and outstanding immediately prior to the Effective Time, shall without further action on the part of the Corporation or any holder of Old Common Stock automatically be reclassified as [range of one-half to one-tenth] of a share of Common Stock. Any stock certificate that, immediately prior to the Effective Time, represented shares of the Old Common Stock will, from an after the Effective Time, automatically and without the necessity of presenting the same for exchange, represent the number of

shares of Common Stock as equals the quotient obtained by dividing the number of shares of Old Common Stock represented by such certificate immediately prior to the Effective Time by [range of two to ten]; provided, however, that each holder of record of a certificate that represented shares of Old Common Stock shall receive upon surrender of such certificate a new certificate representing the number of shares of Common Stock into which the shares of Old Common Stock represented by such certificate have been reclassified pursuant hereto. In all cases, fractional shares resulting from the reclassification will be rounded up to the nearest whole share.

A-1

Table of Contents

6. This Certificate of Amendment shall become effective immediately upon filing with the State of New Jersey.	
	BY: (Signature)
Dated this [] day of [], 20	
May be executed by the Chairman of the Board, or the President, or a Vice President of the Corporation.	
A-2	

EXHIBIT B

EMCORE CORPORATION 2010 EQUITY INCENTIVE PLAN

(as amended and restated effective as of March 2, 2011)

ARTICLE I

PURPOSES

The purposes of the Plan are to foster and promote the long-term financial success of the Company and the Subsidiaries and materially increase shareholder value by (a) motivating superior performance by Participants, (b) providing Participants with an ownership interest in the Company, and (c) enabling the Company and the Subsidiaries to attract and retain the services of outstanding Employees upon whose judgment, interest and special effort the successful conduct of its operations is largely dependent.

ARTICLE II

DEFINITIONS

- 2.1 Certain Definitions. Capitalized terms used herein without definition shall have the respective meanings set forth below:
- "Adjustment Event" means any dividend payable in capital stock, stock split, share combination, extraordinary cash dividend, recapitalization, reorganization, merger, consolidation, split-up, spin-off, combination, exchange of shares or other similar event affecting the Common Stock.
- "Affiliate" means (i) any corporation or limited liability company in an unbroken chain of corporations or limited liability companies ending with the Company if each corporation or limited liability company owns stock or membership interests (as applicable) possessing more than fifty percent (50%) of the total combined voting power of all classes of stock in one of the other corporations or limited liability companies in such chain; (ii) any Subsidiary of the Company or any Affiliate of the Company; or (iii) any other entity, approved by the Committee as an Affiliate under the Plan, in which the Company or any of its Affiliates has a material equity interest.
- "Alternative Award" has the meaning given in Section 8.2.
- "Award" means any Option, Stock Appreciation Right, Performance Stock, Performance Stock Unit, Performance Unit, Restricted Stock, Restricted Stock Unit, Stock Purchase Right, Share Award or Deferred Stock Unit, granted pursuant to the Plan, including an Award combining two or more types in a single grant.
- "Award Agreement" means any written agreement, contract, or other instrument or document evidencing any Award granted by the Committee pursuant to the Plan. The terms of any plan, policy or guideline adopted by the Committee and applicable to an Award shall be deemed incorporated in and part of the related Award Agreement. The Committee may provide for the use of electronic, internet or other non-paper Award Agreements, and the use of electronic, internet or other non-paper means for the Participant's acceptance of, or actions under, an Award Agreement unless otherwise expressly specified herein. In the event of any inconsistency or conflict between the terms of the Plan and an Award Agreement, the terms of the Plan shall govern.

"Beneficial Ownership" (including correlative terms) shall have the same meaning given such term in Rule 13d-3 promulgated under the Exchange Act.

"Board" means the Board of Directors of the Company.

"Cause" means (as determined by the Committee) (i) willful and continued failure to perform substantially the Participant's material duties with the Company (other than any such failure resulting from the Participant's incapacity as a result of physical or mental illness) after a written demand for substantial performance specifying the manner in which the Participant has not performed such duties is delivered to the Participant by the person or entity that supervises or manages the Participant, (ii) engaging in willful and serious misconduct that is injurious to the Company or any of its Subsidiaries, (iii) one or more acts of fraud or personal dishonesty resulting in or intended to result in personal enrichment at the expense of the Company or any of its Subsidiaries, (iv) substantial abusive use of alcohol, drugs or similar substances that, in the sole judgment of the Company, impairs the Participant's job performance, (v) material violation of any Company policy that results in harm to the Company or any of its Subsidiaries or (vi) indictment for or conviction of (or plea of guilty or nolo contendere) to a felony or of any crime (whether or not a felony) involving moral turpitude. A "Termination for Cause," shall include a determination by the Committee following a Participant's termination of employment for any other reason that, prior to such termination of employment, circumstances constituting Cause existed with respect to such Participant.

"Change in Control" means, unless otherwise determined by the Committee at or after the grant date, the occurrence of any of the following:

- (i) an acquisition in one transaction or a series of related transactions (other than directly from the Company or pursuant to Awards granted under the Plan or other similar awards granted by the Company) of any Voting Securities by any Person, immediately after which such Person has Beneficial Ownership of fifty percent (50%) or more of the combined voting power of the Company's then outstanding Voting Securities; provided, however, in determining whether a Change in Control has occurred hereunder, Voting Securities which are acquired in a Non-Control Acquisition shall not constitute an acquisition that would cause a Change in Control;
- (ii) the individuals who, immediately prior to the effective date of this Plan, are members of the Board (the "Incumbent Board"), cease for any reason to constitute at least a majority of the members of the Board; provided, however, that if the election, or nomination for election, by the Company's common stockholders, of any new director was approved by a vote of at least a majority of the Incumbent Board, such new director shall, for purposes of the Plan, be considered as a member of the Incumbent Board; provided further, however, that no individual shall be considered a member of the Incumbent Board if such individual initially assumed office as a result of either an actual or threatened "Election Contest" (as described in Rule 14a-11 promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board (a "Proxy Contest") including by reason of any agreement intended to avoid or settle any Election Contest or Proxy Contest; or
- (iii) the consummation of:
- (A) a merger, consolidation or reorganization involving the Company unless:
- (1) the stockholders of the Company, immediately before such merger, consolidation or reorganization, own, directly or indirectly, immediately following such merger, consolidation or reorganization, more than fifty percent (50%) of the combined voting power of the outstanding voting securities of the corporation resulting from such merger or consolidation or reorganization (the "Surviving Corporation") in substantially the same proportion as their ownership of the Voting Securities immediately before such merger, consolidation or reorganization,
- (2) the individuals who were members of the Incumbent Board immediately prior to the execution of the agreement providing for such merger, consolidation or reorganization constitute at least a majority of the members of the board of directors of the Surviving Corporation, or a corporation Beneficially Owning, directly or indirectly, a majority of

the voting securities of the Surviving Corporation, and

(3) no Person, other than (i) the Company, (ii) any Related Entity, (iii) any employee benefit plan (or any trust forming a part thereof) that, immediately prior to such merger, consolidation or reorganization, was maintained by the Company, the Surviving Corporation, or any Related Entity or (iv) any Person who, together with its Affiliates, immediately prior to such merger, consolidation or reorganization had Beneficial Ownership of fifty percent (50%) or more of the then outstanding Voting Securities, owns, together with its Affiliates, Beneficial Ownership of fifty percent (50%) or more of the combined voting power of the Surviving Corporation's then outstanding voting securities (a transaction described in clauses (1) through (3) above is referred to herein as a "Non-Control Transaction");

(B) a complete liquidation or dissolution of the Company; or

(C) an agreement for the sale or other disposition of all or substantially all of the assets or business of the Company to any Person (other than a transfer to a Related Entity or the distribution to the Company's stockholders of the stock of a Related Entity or any other assets).

Notwithstanding the foregoing, a Change in Control shall not be deemed to occur solely because any Person (the "Subject Person") acquired Beneficial Ownership of fifty percent (50%) or more of the combined voting power of the then outstanding Voting Securities as a result of the acquisition of Voting Securities by the Company which, by reducing the number of Voting Securities then outstanding, increases the proportional number of shares Beneficially Owned by the Subject Persons, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of Voting Securities by the Company, and (1) before such share acquisition by the Company the Subject Person becomes the Beneficial Owner of any new or additional Voting Securities in a related transaction or (2) after such share acquisition by the Company the Subject Person becomes the Beneficial Owner of any new or additional Voting Securities which in either case increases the percentage of the then outstanding Voting Securities Beneficially Owned by the Subject Person, then a Change in Control shall be deemed to occur. Solely for purposes of this Change in Control definition, (x) "Affiliate" shall mean, with respect to any Person, any other Person that, directly or indirectly, controls, is controlled by, or is under common control with, such Person; (y) any "Relative" (for this purpose, "Relative" means a spouse, child, parent, parent of spouse, sibling or grandchild) of an individual shall be deemed to be an Affiliate of such individual for this purpose; and (z) neither the Company nor any Person controlled by the Company shall be deemed to be an Affiliate of any holder of Common Stock.

"Change in Control Price" means the price per share on a fully diluted basis offered in conjunction with any transaction resulting in a Change in Control, as determined in good faith by the Committee as constituted before the Change in Control, if any part of the offered price is payable other than in cash.

"Code" means the Internal Revenue Code of 1986, as amended, and the regulations promulgated thereunder.

"Committee" means the Compensation Committee of the Board or, if applicable, the delegate of the Compensation Committee of the Board as permitted or required herein.

"Common Stock" means the no par value common stock of the Company and any other securities into which the Common Stock is changed or for which the Common Stock is exchanged.

"Company" means EMCORE Corporation, a New Jersey corporation, and any successor thereto.

"Deferred Stock Unit" means a Participant's contractual right to receive a stated number of shares of Common Stock or, if provided by the Committee on or after the grant date, cash equal to the Fair Market Value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to such stated number of shares of Common Stock, under the Plan at the end of a specified period of time.

"Disability" means, unless otherwise provided in an Award Agreement, a physical or mental disability or infirmity that prevents or is reasonably expected to prevent the performance of a Participant's employment-related duties for a period of six months or longer and, within 30 days after the Company notifies the Participant in writing that it intends to terminate his employment, the Participant shall not have returned to the performance of his employment-related duties on a full-time basis; provided, that (i) for purposes of Section 5.3(a) in respect of ISOs, the term "Disability" shall have the meaning assigned to the term "Permanent and Total Disability" by section 22(e)(3) of the Code (i.e., physical or mental disability or infirmity lasting not less than 12 months), and (ii) with respect to any Award that constitutes deferred compensation subject to section 409A of the Code, "Disability" shall have the meaning set forth in section 409A(a)(2)(c) of the Code. The Committee's reasoned and good faith judgment of Disability shall be final, binding and conclusive, and shall be based on such competent medical evidence as shall be presented to it by such Participant and/or by any physician or group of physicians or other competent medical expert employed by the Participant or the Company to advise the Committee. Notwithstanding the foregoing (but except in the case of ISOs and awards subject to section 409A of the Code), with respect to any Participant who is a party to an employment agreement with the Company or any Subsidiary, "Disability" shall have the meaning, if any, specified in such Participant's employment agreement.

"Dividend Equivalents" means an amount equal to any dividends and distributions paid by the Company with respect to the number of shares of Common Stock subject to an Award.

"Employee" means any non-employee director, officer or employee of, or any natural person who is a consultant to, the Company or any Affiliate.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules promulgated thereunder.

"Executive Officer" means each person who is an officer of the Company or any Subsidiary and who is subject to the reporting requirements under Section 16(a) of the Exchange Act.

"Fair Market Value" of a share of Common Stock as of a given date shall be: (i) if the Common Stock is listed or admitted to trading on an established stock exchange (including, for this purpose, The Nasdaq Global Market that comprises part of The Nasdaq Stock Market), the closing sale price for a share of Stock on the composite tape or in Nasdaq Global Market trading as reported in The Wall Street Journal (or, if not so reported, such other nationally recognized reporting source as the Committee shall select) on such date, or, if no such price is reported on such date, the most recent day for which such price is available shall be used; (ii) if the Common Stock is not then listed or admitted to trading on such a stock exchange, the closing sale price for a share of Common Stock on such date as reported by The Nasdaq Capital Market or, if not so reported, by the OTC Bulletin Board (or any successor or similar quotation system regularly reporting the market value of the Common Stock in the over-the-counter market), or, if no such price is reported for such date, the most recent day for which such price is available shall be used; or (iii) in the event neither of the valuation methods provided for in clauses (i) and (ii) above is practicable, the fair market value of a share of Common Stock determined by such other reasonable valuation method as the Committee shall, in its discretion, select and apply in good faith as of the given date; provided, however, that for purposes of an ISO, such fair market value shall be determined subject to section 422(c)(7) of the Code.

"Incumbent Board" has the meaning given in the definition of "Change of Control".

"ISOs" has the meaning given in Section 5.1(a).

"New Employer" means a Participant's employer, or the parent or a subsidiary of such employer, immediately following a Change in Control.

Table of Contents

"NSOs" has the meaning given in Section 5.1(a).

"Non-Control Acquisition" - an acquisition by (i) an employee benefit plan (or a trust forming a part thereof) maintained by (A) the Company or (B) any corporation or other Person of which a majority of its voting power or its voting equity securities or equity interest is owned, directly or indirectly, by the Company (a "Related Entity"), (ii) the Company or any Related Entity, or (iii) any Person in connection with a Non-Control Transaction.

"Non-Control Transaction" has the meaning given in the definition of "Change of Control".

"Option" means the right granted to a Participant pursuant to the Plan to purchase a stated number of shares of Common Stock at a stated price for a specified period of time.

"Participant" means any Employee or prospective Employee designated by the Committee to receive an Award under the Plan.

"Performance Period" means the period, as determined by the Committee, during which the performance of the Company, any Subsidiary, any business unit and any individual is measured to determine whether and the extent to which the applicable performance measures have been achieved, provided that each such period shall be no greater than five years in length.

"Performance Stock" means a grant of a stated number of shares of Common Stock to a Participant under the Plan that is forfeitable by the Participant until the attainment of specified performance goals, or until otherwise determined by the Committee or in accordance with the Plan, subject to the continuous employment of the Participant through the completion of the applicable Performance Period (or such portion of the applicable Performance Period as otherwise provided in Article VI).

"Performance Stock Unit" means a Participant's contractual right to receive a stated number of shares of Common Stock or, if provided by the Committee on or after the grant date, cash equal to the Fair Market Value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to such stated number of shares of Common Stock, under the Plan at a specified time that is forfeitable by the Participant until the attainment of specified performance goals, or until otherwise determined by the Committee or in accordance with the Plan, subject to the continuous employment of the Participant through the completion of the applicable Performance Period (or such portion of the applicable Performance Period as otherwise provided in Article VI).

"Performance Unit" means a Participant's contractual right to receive a cash-denominated award, payable in cash or shares of Common Stock or a combination thereof, under the Plan at a specified time that is forfeitable by the Participant until the attainment of specified performance goals, or until otherwise determined by the Committee or in accordance with the Plan, subject to the continuous employment of the Participant through the applicable Performance Period (or such portion of the applicable Performance Period as otherwise provided in Article VI).

"Permitted Transferee" has the meaning given in Section 11.1.

"Person" means "person" as such term is used for purposes of Section 13(d) or 14(d) of the Exchange Act, including, without limitation, any individual, corporation, limited liability company, partnership, trust, unincorporated organization, government or any agency or political subdivision thereof, or any other entity or any group of Persons.

"Plan" means this EMCORE Corporation 2010 Equity Incentive Plan, as the same may be interpreted by the Committee and/or be amended from time to time.

Table of Contents

"Proxy Contest" has the meaning given in the definition of "Change of Control".

"Related Entity" has the meaning given in the definition of "Non-Control Acquisition".

"Replacement Award" means an Award made to employees of companies or businesses acquired by the Company to replace incentive awards and opportunities held by such employees prior to such acquisition.

"Restricted Stock" means a grant of a stated number of shares of Common Stock to a Participant under the Plan that is forfeitable by the Participant until the completion of a specified period of future service, or until otherwise determined by the Committee or in accordance with the Plan.

"Restricted Stock Unit" means a Participant's contractual right to receive a stated number of shares of Common Stock or, if provided by the Committee on or after the grant date, cash equal to the Fair Market Value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to such stated number of shares of Common Stock, under the Plan at the end of a specified period of time that is forfeitable by the Participant until the completion of a specified period of future service, or until otherwise determined by the Committee or in accordance with the Plan.

"Restriction Period" means (i) with respect to any Performance Stock, Performance Stock Unit or Performance Unit, the period beginning on the grant date of such Award and ending on the certification by the Committee that the performance objectives or objectives for the applicable Performance Period have been attained (in whole or in part) in accordance with Section 6.2(d), (ii) with respect to any Restricted Stock or Restricted Stock Unit, the Restriction Period specified in the Award Agreement evidencing such Award, and (iii) with respect to any freestanding Deferred Stock Unit as to which the Committee has specified a Restriction Period in accordance with Section 7.6, the Restriction Period so specified.

"Retained Award" has the meaning given in Section 6.6(a).

"Retirement" means, except as otherwise defined in an Award Agreement, a Participant's retirement from active employment with the Company and any Subsidiary at or after such Participant attains age 65, or after such Participant attains age 55 and has provided, at minimum, 10 years of service to the Company or any Subsidiary.

"Share Award" means an Award of unrestricted shares of Common Stock pursuant to Section 7.7 of the Plan.

"Stock Appreciation Right" means, with respect to shares of Common Stock, the right to receive a payment from the Company in cash and/or shares of Common Stock equal to the product of (i) the excess, if any, of the Fair Market Value of one share of Common Stock on the exercise date over a specified base price fixed by the Committee on the grant date, multiplied by (ii) a stated number of shares of Common Stock.

"Stock Purchase Right" means the right to purchase Common Stock pursuant to Section 5.5 of the Plan, as evidenced by an Award Agreement.

"Subject Person" has the meaning given in the definition of "Change of Control".

"Subsidiary" means any corporation in which the Company owns, directly or indirectly, stock representing 50% or more of the combined voting power of all classes of stock entitled to vote, and any other business organization, regardless of form, in which the Company possesses, directly or indirectly, 50% or more of the total combined equity interests in such organization.

"Surviving Corporation" has the meaning given in the definition of "Change of Control".

Table of Contents

"Vesting Date" means (i) with respect to any Performance Stock, Performance Stock Unit, Performance Unit, Restricted Stock or Restricted Stock Unit, the expiration date of the applicable Restriction Period, and (ii) with respect to any Option or Stock Appreciation Right, the date such Award first becomes exercisable in accordance with the Plan and the Award Agreement evidencing such Award.

"Voting Securities" means all the outstanding voting securities of the Company entitled to vote generally in the election of the Board.

2.2 Gender and Number. Except when otherwise indicated by the context, words in the masculine gender used in the Plan shall include the feminine gender, the singular shall include the plural, and the plural shall include the singular.

ARTICLE III

POWERS OF THE COMMITTEE

- 3.1 Eligibility and Participation. Participants in the Plan shall be those Employees designated by the Committee (or its delegate) to participate in the Plan.
- 3.2 Power to Grant and Establish Terms of Awards. The Committee shall have the authority, subject to the terms of the Plan, to determine the Employees to whom Awards shall be granted, the type or types of Awards to be granted and the terms and conditions of any and all Awards including, but not limited to, the number of shares of Common Stock subject to an Award, the time or times at which Awards shall be granted, and the terms and conditions of applicable Award Agreements (including, without limitation, where and as permitted under the Plan, terms and conditions that may vary from those under the Plan). The Committee may establish different terms and conditions for different types of Awards, for different Participants receiving the same type of Award, and for the same Participant for each type of Award such Participant may receive, whether or not granted at the same or different times.
- 3.3 Administration. The Committee shall be responsible for the administration of the Plan. Any Awards granted by the Committee may be subject to such conditions, not inconsistent with the terms of the Plan, as the Committee shall determine. The Committee shall have authority to prescribe, amend and rescind rules and regulations relating to the Plan, to provide for conditions deemed necessary or advisable to protect the interests of the Company, to interpret the Plan and to make all other determinations necessary or advisable for the administration and interpretation of the Plan and to carry out its provisions and purposes. Any determination, interpretation or other action made or taken (including any failure to make any determination or interpretation, or take any other action) by the Committee pursuant to the provisions of the Plan, shall, to the greatest extent permitted by law, be within its sole and absolute discretion and shall be final, binding and conclusive for all purposes and upon all persons and shall be given deference in any proceeding with respect thereto. The Committee may appoint accountants, actuaries, counsel, advisors and other persons that it deems necessary or desirable in connection with the administration of the Plan. The Committee's determinations under the Plan need not be uniform and may be made by the Committee selectively among persons who receive, or are eligible to receive, Awards under the Plan, whether or not such persons are similarly situated. To the maximum extent permitted by law, no member of the Committee shall be liable for any action taken or decision made in good faith relating to the Plan or any Award hereunder.
- 3.4 Delegation by the Committee. The Committee may delegate, subject to the provisions of this Plan, EMCORE Corporation's Incentive Stock Option Grant Policy and such other policies as the Committee may adopt, to any officer or group of officers, or director or group of directors of the Company (including to a subcommittee of members of the Compensation Committee of the Board) or its affiliates any portion of its authority and powers under the Plan with respect to Participants who are not Executive Officers; provided, that any delegation to one or more

officers of the Company shall be subject to N.J.S.A. Section14A:8-1(4) (or successor provision). Only the Committee may select, grant, administer, or exercise any other discretionary authority under the Plan in respect of Awards granted to such Participants who are Executive Officers. Notwithstanding the foregoing, (i) with respect to any Award intended to qualify as "performance-based" compensation under section 162(m) of the Code, the Committee shall consist solely of two or more "outside directors" within the meaning of the regulations promulgated under section 162(m) of the Code, and (ii) with respect to any award intended to qualify for the exemption contained in Rule 16b-3 promulgated under the Exchange Act, the Committee shall consist solely of two or more "non-employee directors" within the meaning of such Rule, or, in the alternative, of the entire Board.

Participants Based Outside the United States. In order to conform with provisions of local laws and regulations in foreign countries in which the Company or its Subsidiaries operate, the Committee may (i) modify the terms and conditions of Awards granted to Participants employed outside the United States, (ii) establish subplans with modified exercise procedures and such other modifications as may be necessary or advisable under the circumstances presented by local laws and regulations, and (iii) take any action which it deems advisable to obtain, comply with or otherwise reflect any necessary governmental regulatory procedures, exemptions or approvals with respect to the Plan or any subplan established hereunder; provided, however, that the Committee may not make any sub-plan that (a) increases the limitations contained in Section 4.3, (b) increases the number of shares available under the Plan, as set forth in Section 4.1; or (c) causes the Plan to cease to satisfy any conditions under Rule 16b-3 under the Exchange Act or causes the grant of any performance Award to fail to qualify for an income tax deduction pursuant to section 162(m) of the Code. Subject to the foregoing, the Committee may amend, modify, administer or terminate such sub-plans, and prescribe, amend and rescind rules and regulations relating to such sub-plans.

ARTICLE IV

STOCK SUBJECT TO PLAN

- Number. Subject to the provisions of this Article IV, the maximum number of shares of Common Stock available for Awards under the Plan shall not exceed 4,000,000 shares of Common Stock (all of which may be the subject of ISOs granted under the Plan); provided, however, that if the Company's shareholders so approve at the 2011 shareholder meeting, the maximum number of shares of Common Stock available for Awards under the Plan shall not exceed 7,000,000 shares of Common Stock (all of which may be the subject of ISOs granted under the Plan). The shares of Common Stock to be delivered under the Plan may consist, in whole or in part, of Common Stock held in treasury or authorized but unissued shares of Common Stock, not reserved for any other purpose.
- 4.2 Canceled, Terminated, or Forfeited Awards, etc. Shares subject to any Award granted under the Plan (other than Replacement Awards) that for any reason are canceled, terminated, forfeited, settled in cash or otherwise settled without the issuance of Common Stock after the effective date of the Plan shall be available for grant under the Plan. Replacement Awards that for any reason are canceled, terminated, forfeited, settled in cash or otherwise settled without the issuance of Common Stock after the effective date of the Plan shall not be available for grant under the Plan. Without limiting the generality of Section 4.1 hereof, (i) shares of Common Stock tendered by a Participant or withheld by the Company to pay the exercise price of any Options, or to satisfy any tax withholding obligations pursuant to Section 10.4, shall be available for grant under the Plan, (ii) upon settlement of Stock Appreciation Rights, a number of shares of Common Stock equal to (x) the number of shares subject to the Stock Appreciation Rights minus (y) that number of shares delivered to the Participant, shall again be available for grant under the Plan, and (iii) shares of Common Stock issued in connection with Awards that are assumed, converted or substituted pursuant to an Adjustment Event or Change in Control (i.e., Alternative Awards), or issued in connection with Replacement Awards, shall not be counted against the maximum limitation specified in Section 4.1. For purposes of this Article IV, if a Stock Appreciation Right is granted in tandem with an Option so that only one may be exercised with the other being surrendered on such exercise in accordance with Section 5.2(b), the number of shares subject to the tandem Option and Stock Appreciation Right award shall only be taken into account once (and not as to both Awards).

Table of Contents

- 4.3 Individual Award Limitations. Subject to the provisions of Sections 4.2 and 4.4, the following individual Award limits shall apply:
- (a) During any 12-month period, no Participant shall receive Options, Stock Appreciation Rights and/or Stock Purchase Rights covering more than 500,000 shares of Common Stock; and
- (b) During any 12-month period, no Participant shall receive any awards of Performance Stock, Performance Stock Units, Restricted Stock, Restricted Stock Units, Deferred Stock Units and/or Share Awards covering more than 200,000 shares of Common Stock.
- (c) During any calendar year, the maximum dollar amount of cash which may be earned in connection with the grant of Performance Units may not exceed \$[_____].
- Adjustment in Capitalization. In the event of any Adjustment Event affecting the Common Stock, the Committee shall make an equitable and proportionate anti-dilution adjustment to offset any resultant change in the per-share price of the Common Stock and preserve the intrinsic value of Options and any other Awards granted under the Plan. Such mandatory adjustment may include a change in any or all of (a) the number and kind of shares of Common Stock which thereafter may be awarded or optioned and sold under the Plan (including, but not limited to, adjusting any limits on the number and types of Awards that may be made under the Plan), (b) the number and kind of shares of Common Stock subject to outstanding Awards, and (c) the grant, exercise or conversion price with respect to any Award. In addition, the Committee may make provisions for a cash payment to a Participant or a person who has an outstanding Award. The number of shares of Common Stock subject to any Award shall be rounded to the nearest whole number. Any such adjustment shall be consistent with sections 424, 409A and 162(m) of the Code to the extent the Awards subject to adjustment are subject to such sections of the Code.
- 4.5 Prohibition Against Repricing. Except to the extent (i) approved in advance by a majority of the shares of the Company entitled to vote generally in the election of directors or (ii) as a result of any Adjustment Event, the Committee shall not have the power or authority to reduce, whether through amendment or otherwise, the exercise price of any outstanding Option or base price of any outstanding Stock Appreciation Right or to grant any new Award, or make any cash payment, in substitution for or upon the cancellation of Options or Stock Appreciation Rights previously granted.

ARTICLE V

STOCK OPTIONS, STOCK APPRECIATION RIGHTS AND STOCK PURCHASE RIGHTS

5.1 Options.

(a) Grant. Options may be granted to Participants at such time or times as shall be determined by the Committee. Options pursuant to this Plan may be of two types: (i) "incentive stock options" within the meaning of section 422 of the Code ("ISOs") and (ii) non-statutory stock options ("NSOs"), which are not ISOs. The grant date of an Option under the Plan will be: (i) with respect to awards to new hires made in accordance with policies and guidelines, the date of commencement of employment (or such future date as the Committee may determine); and (ii) with respect to all other awards, the date on which the Option is approved by the Committee or such other future date as the Committee shall determine. Each Option shall be evidenced by an Award Agreement that shall specify the type of Option granted, the exercise price, the duration of the Option, the number of shares of Common Stock to which the Option pertains, and such other conditions as the Committee shall determine, including customary representations, warranties and covenants with respect to securities law matters. Each Option shall be an ISO unless otherwise designated by the Committee at the time of grant or in the Award Agreement evidencing such Option, or unless such

Option does not otherwise meet the requirements of section 422 of the Code.

- (b) Exercise Price. Each Option granted pursuant to the Plan shall have an exercise price per share of Common Stock determined by the Committee; provided, that except in the case of Replacement Awards, such per share exercise price may not be less than the Fair Market Value of one share of Common Stock on the Option grant date.
- (c) Exercisability. Each Option awarded to a Participant under the Plan shall become exercisable based on the performance of a minimum period of service or the occurrence of any event or events, including a Change in Control, as the Committee shall determine, either at or after the grant date. No Option shall be exercisable on or after the tenth anniversary of its grant date. Except as otherwise provided in the Plan, the applicable Award Agreement or as determined by the Committee at or after the grant date, after becoming exercisable each installment of an Option shall remain exercisable until expiration, termination or cancellation of the Option and, until such time, may be exercised from time to time in whole or in part, up to the total number of shares of Common Stock with respect to which it is then exercisable.
- Payment. The Committee shall establish procedures governing the exercise of Options, which procedures shall generally require that written notice of exercise thereof be given and that the exercise price thereof and any applicable withholding tax obligations be paid in full at the time of exercise (i) in cash or cash equivalents, including by personal check, (ii) through delivery of shares of Common Stock (either in full or in part, and including actual delivery or delivery by attestation), including, but not limited to, the election by the Participant to reduce the number of shares of Common Stock that are subject to the portion of the Options being exercised having a Fair Market Value equal to such portion, or (iii) in accordance with such other procedures or in such other forms as the Committee shall from time to time determine, which may include a broker-assisted cashless exercise arrangement.

5.2 Stock Appreciation Rights.

- (a) Grant. Stock Appreciation Rights may be granted to Participants at such time or times as shall be determined by the Committee. Stock Appreciation Rights may be granted in tandem with Options which, unless otherwise determined by the Committee at or after the grant date, shall have substantially similar terms and conditions to such Options to the extent applicable, or may granted on a freestanding basis, not related to any Option. The grant date of any Stock Appreciation Right under the Plan will be the date on which the Stock Appreciation Right is approved by the Committee or such other future date as the Committee shall determine. No Stock Appreciation Right shall be exercisable on or after the tenth anniversary of its grant date. Stock Appreciation Rights shall be evidenced in writing, whether as part of the Award Agreement governing the terms of the Options, if any, to which such Stock Appreciation Rights, in each case, containing such conditions as the Committee shall determine, including customary representations, warranties and covenants with respect to securities law matters.
- (b) Exercise. Stock Appreciation Rights awarded to a Participant under the Plan shall become exercisable based on the performance of a minimum period of service or the occurrence of any event or events, including a Change in Control, as the Committee shall determine, either at or after the grant date. Stock Appreciation Rights that are granted in tandem with an Option may only be exercised upon the surrender of the right to exercise such Option for an equivalent number of shares of Common Stock, and may be exercised only with respect to the shares of Common Stock for which the related Option is then exercisable.
- (c) Settlement. Upon exercise of a Stock Appreciation Right, the Participant shall be entitled to receive payment in the form, determined by the Committee, of cash or shares of Common Stock having a Fair Market Value equal to such cash amount, or any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to such cash amount, determined by multiplying: (i) any increase in the Fair Market Value of one share of Common Stock on the exercise date over the base price fixed by the Committee on the grant date of such Stock Appreciation Right, which may not be less than the Fair Market Value of a share of Common Stock on the grant date

of such Stock Appreciation Right (except if awarded in tandem with an Option but after the grant date of such Option, then not less than the exercise price of such Option), by (ii) the number of shares of Common Stock with respect to which the Stock Appreciation Right is exercised. Notwithstanding the foregoing, on the grant date the Committee may establish a maximum amount per share which will be payable upon exercise of a Stock Appreciation Right.

- 5.3 Termination of Employment.
- (a) Death or Disability. Unless otherwise determined by the Committee at or after the grant date, if a Participant's employment terminates by reason of such Participant's death or Disability, any Options and Stock Appreciation Rights granted to such Participant, whether or not exercisable on or prior to the date of such termination, shall, subject to Section 8.1, be exercisable by the Participant (or the Participant's designated beneficiary, as applicable) at any time prior to the second anniversary of the Participant's termination of employment or the expiration of the term of the Options and Stock Appreciation Rights, whichever period is shorter, and thereafter any Options and Stock Appreciation Rights that have not been exercised shall be forfeited and canceled.
- (b) For Cause. If a Participant's employment with the Company or any Subsidiary is terminated for Cause, all Options and Stock Appreciation Rights granted to such Participant which are then outstanding (whether or not exercisable on or prior to the date of such termination) shall be immediately forfeited and canceled.
- Any Other Reason. Unless otherwise determined by the Committee at or after the grant date, if a Participant's employment is terminated for any reason other than the ones described in Section 5.3(a) or (b) the Participant may exercise any Options and Stock Appreciation Rights that are exercisable on the date of such termination until the earlier of (i) the 90th day following the date of such termination or, if later, the 90th day following expiration of any blackout period in effect with respect to such Options and Stock Appreciation Rights, and (ii) the expiration of the term of such Options and Stock Appreciation Rights. Any Options and Stock Appreciation Rights that are not exercisable upon termination of a Participant's employment shall be forfeited and canceled as of the date of such termination.
- Committee Discretion. Notwithstanding anything to the contrary contained in this Article V, the Committee may, at or after the date of grant, accelerate or waive any conditions to the exercisability of any Option or Stock Appreciation Right granted under the Plan, and may permit all or any portion of any such Option or Stock Appreciation Right to be exercised following a Participant's termination of employment for any reason on such terms and subject to such conditions as the Board shall determine for a period up to and including, but not beyond, the expiration of the term of such Options or Stock Appreciation Rights.
- 5.5 Stock Purchase Rights.
- (a) Grant. Stock Purchase Rights may be granted to Participants at such time or times as shall be determined by the Committee. Each Stock Purchase Right shall be evidenced by an Award Agreement that shall specify the terms, conditions and restrictions related to the offer, including the number of shares of Common Stock that the offeree shall be entitled to purchase, the price to be paid, the time within which the offeree must accept such offer, and such other conditions as the Committee shall determine, including customary representations, warranties and covenants with respect to securities law matters. The offer shall be accepted by execution of the Award Agreement.
- (b) Rights to Purchase. Stock Purchase Rights may be issued either alone, in addition to, or in tandem with other Awards granted under the Plan and/or cash awards made outside of the Plan.
- (c) Repurchase Option. Unless the Committee determines otherwise, the Award Agreement shall grant the Company a repurchase option exercisable upon the voluntary or involuntary termination of the purchaser's service with the Company (or any Subsidiary) for any reason (including death or Disability). The purchase price for shares of Common Stock repurchased pursuant to the Award Agreement shall be the original price paid by the purchaser and may be paid by cancellation of any indebtedness of the purchaser to the Company. The repurchase option shall lapse at a rate determined by the Committee.

- Other Provisions. The Committee may set restrictions based upon the achievement of specific performance objectives (Company-wide, divisional, or individual), applicable federal or state securities laws, or any other basis determined by the Committee in its discretion. For purposes of qualifying grants of Stock Purchase Rights as "performance-based compensation" under section 162(m) of the Code, the Committee, in its discretion, may set restrictions based upon the achievement of performance objectives as described in Sections 6.2(b)-(d). The performance objectives will be set by the Committee on or before the latest date permissible to enable the Stock Purchase Rights to qualify as "performance-based compensation" under section 162(m) of the Code. In granting Stock Purchase Rights which are intended to qualify under section 162(m) of the Code, the Committee will follow any procedures determined by it from time to time to be necessary or appropriate to ensure qualification of the Stock Purchase Rights under section 162(m) of the Code (e.g., in determining the performance objectives).
- (e) Rights as a Stockholder. Once the Stock Purchase Right is exercised, the purchaser shall have the rights equivalent to those of a stockholder, and shall be a stockholder when his or her purchase is entered upon the records of the duly authorized transfer agent of the Company. No adjustment will be made for a dividend or other right for which the record date is prior to the date the Stock Purchase Right is exercised, except as otherwise specifically provided under the Plan.

ARTICLE VI

PERFORMANCE STOCK, PERFORMANCE STOCK UNITS AND PERFORMANCE UNITS

Grant. Performance Stock, Performance Stock Units and Performance Units may be granted to Participants at such time or times as shall be determined by the Committee. The grant date of any Performance Stock, Performance Stock Units and Performance Units under the Plan will be the date on which such Performance Stock, Performance Stock Units and Performance Units are awarded by the Committee or on such other future date as the Committee shall determine. Performance Stock, Performance Stock Units and Performance Units shall be evidenced by an Award Agreement that shall specify the number of shares of Performance Stock, the number of Performance Stock Units, or the dollar amount of any Performance Units, as the case may be, to which such Award pertains, the Restriction Period, the Performance Period, and such other conditions as the Committee shall determine, including customary representations, warranties and covenants with respect to securities law matters. No shares of Common Stock will be issued at the time an Award of Performance Stock Units or Performance Units is made, and the Company shall not be required to set aside a fund for the payment of any such Award.

6.2 Vesting.

In General. Performance Stock, Performance Stock Units and Performance Units granted to a Participant under the Plan shall be subject to a Restriction Period, which shall lapse upon the attainment of specified performance objectives or the occurrence of any event or events, including a Change in Control, as the Committee shall determine, either at or after the grant date. The Committee shall establish the performance objectives upon which the Restriction Period shall lapse, which, in the case of any such Award intended to qualify as "performance-based" compensation under section 162(m) of the Code, shall be established no later than the 90th day after the applicable Performance Period begins (or such other date as may be required or permitted under section 162(m) of the Code).

- Performance Objectives. The performance objectives for any grant of Performance Stock, Performance (b) Stock Units and Performance Units intended to qualify as "performance-based" compensation under section 162(m) of the Code will be based upon the relative or comparative achievement of one or more of the following criteria, as determined by the Committee: net sales; revenue; revenue growth or product revenue growth; operating income (before or after taxes); pre- or after-tax income (before or after allocation of corporate overhead and bonus); net earnings; earnings per share; net income (before or after taxes); return on equity; total shareholder return; return on assets or net assets; appreciation in and/or maintenance of share price; market share; gross profits; earnings (including adjusted pre-tax earnings, earnings before taxes, earnings before interest and taxes or earnings before interest, taxes, depreciation and amortization); economic value-added models or equivalent metrics; comparisons with various stock market indices; reductions in costs; total net cash flow; cash flow or cash flow per share (before or after dividends); return on capital (including return on total capital or return on invested capital); cash flow return on investment; improvement in or attainment of expense levels or working capital levels; operating margins, gross margins or cash margin; year-end cash; debt reductions; shareholder equity; market share; regulatory achievements; and implementation, completion or attainment of measurable objectives with respect to customer satisfaction, research, development, products or projects and recruiting and maintaining personnel. The performance objectives for any grant of Performance Stock, Performance Stock Units and Performance Units not intended to qualify as "performance-based" compensation under section 162(m) of the Code will be based on the foregoing or such other criteria as may be determined by the Committee.
- Company-wide basis or with respect to one or more Company business units or divisions, or Subsidiaries; and either in absolute terms, relative to the performance of one or more similarly situated companies, or relative to the performance of an index covering a peer group of companies. When establishing performance objectives for the applicable Performance Period, the Committee may exclude any or all "extraordinary items" as determined under U.S. generally acceptable accounting principles including, without limitation, the charges or costs associated with restructurings of the Company, discontinued operations, other unusual or non-recurring items, and the cumulative effects of accounting changes, and as identified in the Company's financial statements, notes to the Company's financial statements or management's discussion and analysis of financial condition and results of operations contained in the Company's most recent report filed with the U.S. Securities and Exchange Commission pursuant to the Exchange Act; provided, that the Committee shall have no discretion with respect to any Award intended to qualify as "performance-based" compensation under section 162(m) of the Code if the exercise of such discretion or the ability to exercise such discretion would cause such Award to fail to qualify as "performance-based" compensation under section 162(m) of the Code.
- Certification of Attainment of Performance Objectives. The Restriction Period with respect to any Performance Stock, Performance Stock Units and Performance Units intended to qualify as "performance-based" compensation under section 162(m) of the Code shall lapse upon the written certification by the Committee that the performance objective or objectives for the applicable Performance Period have been attained. The Committee may provide at the time of grant that if the performance objective or objectives are attained in part, the Restriction Period with respect to a specified portion (which may be zero) of the any Performance Stock, Performance Stock Units and Performance Units will lapse and any remaining portion shall be cancelled; provided, that the Committee shall have no discretion to take such action with respect to any Award intended to qualify as "performance-based" compensation under section 162(m) of the Code if the exercise of such action or the ability to exercise such action would cause such Award to fail to qualify as "performance-based" compensation under section 162(m) of the Code.
- (e) Newly Eligible Participants. Notwithstanding anything in this Article VI to the contrary, the Committee shall be entitled to make such rules, determinations and adjustments as it deems appropriate with respect to any Participant who becomes eligible to receive an Award of Performance Stock, Performance Stock Units and Performance Units after the commencement of a Performance Period.

- 6.3 Additional Provisions Relating to Performance Stock.
- Restrictions on Transferability. Except as otherwise provided in Section 6.6(a), no Performance Stock may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated until the lapse of the Restriction Period. Thereafter, Performance Stock may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated in compliance with all applicable securities laws, the Award Agreement and any other agreement to which the Performance Stock is subject. The Committee shall require that any stock certificates evidencing any Performance Stock be held in the custody of the Secretary of the Company until the applicable Restriction Period lapses, and that, as a condition of any grant of Performance Stock, the Participant shall have delivered a stock power, endorsed in blank, relating to the shares of Common Stock covered by such Award. Any attempt by a Participant, directly or indirectly, to offer, transfer, sell, pledge, hypothecate or otherwise dispose of any Performance Stock or any interest therein or any rights relating thereto without complying with the provisions of the Plan, including this Section 6.3, shall be void and of no effect.
- (b) Legend. Each certificate evidencing shares of Common Stock subject to an Award of Performance Stock shall be registered in the name of the Participant holding such Performance Stock and shall bear the following (or similar) legend:

"THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE TERMS AND CONDITIONS (INCLUDING FORFEITURE) CONTAINED IN THE EMCORE CORPORATION 2010 EQUITY INCENTIVE PLAN AND THE RELATED AWARD AGREEMENT AND NEITHER THIS CERTIFICATE NOR THE SHARES REPRESENTED BY IT ARE ASSIGNABLE OR OTHERWISE TRANSFERABLE EXCEPT IN ACCORDANCE WITH SUCH PLAN, A COPY OF WHICH IS ON FILE WITH THE SECRETARY OF THE COMPANY."

- Rights as a Stockholder. The Committee shall determine whether and to what extent dividends and distributions will be credited to the account of a Participant receiving an Award of Performance Stock. Unless otherwise determined by the Committee at or after the grant date, (i) any cash dividends or distributions credited to the Participant's account shall be deemed to have been invested in additional Performance Stock on the payment date established for the related dividend or distribution in an amount per share of Performance Stock equal to the greatest whole number which may be obtained by dividing (A) the value of such dividend or distribution on the record date by (B) the Fair Market Value of one share of Common Stock on such date, and any such additional Performance Stock shall be subject to the same terms and conditions as are applicable in respect of the Performance Stock with respect to which such dividends or distributions were payable, and, (ii) if any such dividends or distributions are paid in shares of Common Stock or other securities, such shares and other securities shall be subject to the same Restriction Period and other restrictions as apply to the Performance Stock with respect to which they were paid. A Participant holding outstanding Performance Stock shall be entitled to exercise full voting rights and other rights as a stockholder with respect to the shares of Common Stock underlying such Award during the period in which such shares remain subject to the Restriction Period.
- 6.4 Additional Provisions Relating to Performance Stock Units.
- (a) Restrictions on Transferability. Except as otherwise provided in Section 6.6(a) or with the consent of the Committee, Performance Stock Units may not be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated. Any attempt by a Participant, directly or indirectly, to offer, transfer, sell, pledge, hypothecate or otherwise dispose of any Performance Stock Units or any interest therein or any rights relating thereto other than as provided in the Plan shall be void and of no effect.
- (b) Rights as a Stockholder. The Committee shall determine whether and to what extent Dividend Equivalents will be credited to the account of a Participant receiving an Award of Performance Stock Units. Unless otherwise

determined by the Committee at or after the grant date, (i) any cash dividends or distributions credited to the Participant's account shall be deemed to have been invested in additional Performance Stock Units on the payment date established for the related dividend or distribution in an amount per Performance Stock Unit equal to the greatest whole number which may be obtained by dividing (A) the value of such dividend or distribution on the record date by (B) the Fair Market Value of one share of Common Stock on such date, and any such additional Performance Stock Units shall be subject to the same terms and conditions as are applicable in respect of the Performance Stock Units with respect to which such dividends or distributions were payable, and (ii) if any such dividends or distributions are paid in shares of Common Stock or other securities, such shares of Common Stock and other securities shall be subject to the same Restriction Period and Performance Period and other restrictions as apply to the Performance Stock Units with respect to which they were paid. Unless and until the Company issues a certificate or certificates to a Participant for shares of Common Stock in respect of his or her Award of Performance Stock Units, or otherwise determined by the Committee at or after the grant date, a Participant holding outstanding Performance Stock Units shall not be entitled to exercise any voting rights and any other rights as a stockholder with respect to the shares of Common Stock underlying such Award.

(c) Settlement of Performance Stock Units. Unless the Committee determines otherwise at or after the grant date, as soon as reasonably practicable after the lapse of the Restriction Period with respect to any Performance Stock Units then held by a Participant, the Company shall issue to the Participant the shares of Common Stock underlying such Performance Stock Units (plus additional shares of Common Stock for each Performance Stock Units credited in respect of dividends or distributions) or, if the Committee so determines in its sole discretion, an amount in cash equal to the Fair Market Value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to such shares of Common Stock. To the extent permitted by applicable law (including section 409A of the Code), upon such terms and conditions as the Committee may establish from time to time, a Participant may be permitted to defer the receipt of the shares of Common Stock or cash otherwise deliverable upon settlement of Performance Stock Units. Upon issuance of shares of Common Stock underlying Performance Stock Units following lapse of the Restriction Period, such shares may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated in compliance with all applicable securities laws, the Award Agreement and any other agreement to which such shares are subject.

6.5 Additional Provisions Relating to Performance Units.

- (a) Restrictions on Transferability. Except as otherwise provided in Section 6.6(a) no Performance Units may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated. Any attempt by a Participant, directly or indirectly, to offer, transfer, sell, pledge, hypothecate or otherwise dispose of any Performance Units or any interest therein or any rights relating thereto shall be void and of no effect.
- (b) Settlement of Performance Units. Unless the Committee determines otherwise at or after the grant date, as soon as reasonably practicable after the lapse of the Restriction Period with respect to any Performance Units then held by a Participant, the Company shall deliver to the Participant a cash payment equal to the value of such Award or, if the Committee has so determined, a number of shares of Common Stock, which shares shall have a Fair Market Value equal to the value of such Award, or any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to the value of such Award. To the extent permitted by applicable law (including section 409A of the Code), upon such terms and conditions as the Committee may establish from time to time, a Participant may be permitted to defer the receipt of cash or the shares of Common Stock otherwise deliverable upon settlement of Performance Units. Upon issuance of shares of Common Stock underlying Performance Units following lapse of the Restriction Period, such shares may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated in compliance with all applicable securities laws, the Award Agreement and any other agreement to which such shares are subject.

6.6 Termination of Employment.

Death or Disability. Unless otherwise determined by the Committee at or after the grant date, if a Participant's employment terminates by reason of such Participant's death or Disability, the Participant or, as the case may be, the Participant's estate, shall retain a portion of his or her Performance Stock, Performance Stock Units and Performance Units equal to the number of shares or units underlying each Award multiplied by a fraction, the numerator of which is the number of days elapsed from the commencement of the applicable Performance Period through the date of termination, and the denominator of which is the number of days in such Performance Period (each a "Retained Award"), and the remainder of each Award shall be forfeited and canceled as of the date of such termination. The Restriction Period on a Retained Award shall lapse upon completion of the applicable Performance Period to the extent that applicable performance objectives are attained. Settlement of a Retained Award shall be made at the time and in the manner provided in Sections 6.4(c) and 6.5(b) except that no additional deferrals shall be permitted.

- (b) Retirement. Unless otherwise determined by the Committee at or after the grant date, if a Participant's employment terminates as a result of his or her Retirement, any Performance Stock, Performance Stock Units and Performance Units for which the Performance Period has not then lapsed shall be forfeited and canceled as of the date of such termination of employment.
- (c) Any Other Reason. Unless otherwise determined by the Committee at or after the grant date, if a Participant's employment is terminated for any reason other than one described in Sections 6.6(a) and (b), the then-outstanding Performance Stock, Performance Stock Units and Performance Units granted to such Participant shall be immediately forfeited and canceled as of the date of such termination of employment.

ARTICLE VII

RESTRICTED STOCK, RESTRICTED STOCK UNITS, DEFERRED STOCK UNITS AND SHARE AWARDS

- Grant of Restricted Stock and Restricted Stock Units. Restricted Stock and Restricted Stock Units may be granted to Participants at such time or times as shall be determined by the Committee. The grant date of any Restricted Stock and Restricted Stock Units under the Plan will be the date on which such Restricted Stock and Restricted Stock Units are awarded by the Committee or on such other future date as the Committee shall determine. Restricted Stock and Restricted Stock Units shall be evidenced by an Award Agreement that shall specify the number of shares of Common Stock to which the Restricted Stock and Restricted Stock Units pertain (and, if applicable, whether such Award may be payable in cash), the Restriction Period, and such terms and conditions as the Committee shall determine, including customary representations, warranties and covenants with respect to securities law matters. No shares of Common Stock will be issued at the time an Award of Restricted Stock Units is made and the Company shall not be required to set aside a fund for the payment of any such Award.
- 7.2 Vesting for Restricted Stock and Restricted Stock Units. Restricted Stock and Restricted Stock Units granted to a Participant under the Plan shall be subject to a Restriction Period, which shall lapse upon the performance of a minimum period of service, or the occurrence of any event or events, including a Change in Control, as the Committee shall determine, either at or after the grant date. The Restriction Period on any Restricted Stock and Restricted Stock Units shall not fully lapse prior to a Participant's completion of three years of service to the Company or any Subsidiary from the date of the Award grant; provided, that the Committee may provide for a Restriction Period to lapse in pro rata or graded installments over such three-year period; provided further, that the Committee may grant Awards for Restricted Stock and Restricted Stock Units for an aggregate number of shares of Common Stock not to exceed 350,000 shares of Common Stock that have a Restriction Period which lapses in full prior to a Participant's completion of three years of service to the Company or any Subsidiary from the date of the Award grant (provided, that, shares subject to any Restricted Stock Award that are forfeited and subsequently made available for grant under the Plan in accordance with the first sentence of Section 4.2 shall not count toward the preceding limitation following their forfeiture); and provided further, that the minimum Restriction Period for any such Award granted to a newly eligible individual or a Replacement Award shall instead be one year, and the minimum Restriction Period for any such Award to a non-employee director of the Company or its Subsidiaries shall be as determined by the Committee.

- 7.3 Additional Provisions Relating to Restricted Stock.
- Restrictions on Transferability. Unless otherwise determined by the Committee, no Restricted Stock may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated until the lapse of the Restriction Period. Thereafter, Restricted Stock may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated in compliance with all applicable securities laws, the Award Agreement, and any other agreement to which the Restricted Stock is subject. The Committee shall require that any stock certificates evidencing any Restricted Stock be held in the custody of the Secretary of the Company until the applicable Restriction Period lapses, and that, as a condition of any grant of Restricted Stock, the Participant shall have delivered a stock power, endorsed in blank, relating to the share covered by such Award. Any attempt by a Participant, directly or indirectly, to offer, transfer, sell, pledge, hypothecate or otherwise dispose of any Restricted Stock or any interest therein or any rights relating thereto without complying with the provisions of the Plan, including this Section 7.3, shall be void and of no effect.
- (b) Legend. Each certificate evidencing shares of Common Stock subject to an Award of Restricted Stock shall be registered in the name of the Participant holding such Restricted Stock and shall bear the legend (or similar legend) as specified in Section 6.3(b).
- Rights as a Stockholder. Unless otherwise determined by the Committee at or after the grant date, a Participant holding outstanding Restricted Stock shall be entitled to (i) receive all dividends and distributions paid in respect of shares of Common Stock underlying such Award; provided, that, if any such dividends or distributions are paid in shares of Common Stock or other securities, such shares and other securities shall be subject to the same Restriction Period and other restrictions as apply to the Restricted Stock with respect to which they were paid, and (ii) exercise full voting rights and other rights as a stockholder with respect to the shares of Common Stock underlying such Award during the period in which such shares remain subject to the Restriction Period.
- 7.4 Additional Provisions Relating to Restricted Stock Units.
- (a) Restrictions on Transferability. No Restricted Stock Units may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated. Any attempt by a Participant, directly or indirectly, to offer, transfer, sell, pledge, hypothecate or otherwise dispose of any Restricted Stock Units or any interest therein or any rights relating thereto without complying with the provisions of the Plan, including this Section 7.4, shall be void and of no effect.
- Rights as a Stockholder. The Committee shall determine whether and to what extent Dividend Equivalents (b) will be credited to the account of, or will be paid currently to, a Participant receiving an Award of Restricted Stock Units. Unless otherwise determined by the Committee at or after the grant date, (i) any cash dividends or distributions credited to the Participant's account shall be deemed to have been invested in additional Restricted Stock Units on the payment date established for the related dividend or distribution in an amount equal to the greatest whole number which may be obtained by dividing (A) the value of such dividend or distribution on the record date by (B) the Fair Market Value of one share of Common Stock on such date, and any such additional Restricted Stock Units shall be subject to the same terms and conditions as are applicable in respect of the Restricted Stock Units with respect to which such dividends or distributions were payable, and (ii) if any such dividends or distributions are paid in shares of Common Stock or other securities, such shares and other securities shall be subject to the same Restriction Period and other restrictions as apply to the Restricted Stock Units with respect to which they were paid. Unless and until the Company issues a certificate or certificates to a Participant for shares of Common Stock in respect of his or her Award of Restricted Stock Units, or otherwise determined by the Committee at or after the grant date, a Participant holding outstanding Restricted Stock Units shall not be entitled to exercise any voting rights and any other rights as a stockholder with respect to the shares of Common Stock underlying such Award.

(c) Settlement of Restricted Stock Units. Unless the Committee determines otherwise at or after the grant date, as soon as reasonably practicable after the lapse of the Restriction Period with respect to any Restricted Stock Units, the Company shall issue the shares of Common Stock underlying such Restricted Stock Units (plus additional shares of Common Stock for Restricted Stock Unit credited in respect of dividends or distributions) or, if the Committee so determines in its sole discretion, an amount in cash equal to the Fair Market Value of such shares of Common Stock or any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to such shares of Common Stock. To the extent permitted by applicable law (including section 409A of the Code), upon such terms and conditions as the Committee may establish from time to time, a Participant may be permitted to defer the receipt of the shares of Common Stock or cash otherwise deliverable upon settlement of Restricted Stock Units. Upon issuance of shares of Common Stock following lapse of the Restriction Period, such shares may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated in compliance with all applicable securities law, the Award Agreement and any other agreement to which such shares are subject.

7.5 Termination of Employment.

- (a) Death or Disability. Unless otherwise determined by the Committee at or after the grant date, if a Participant's employment terminates by reason of such Participant's death or Disability, the Participant or, as the case may be, the Participant's estate, shall retain a portion of his or her Restricted Stock and Restricted Stock Units equal to the number of shares or units underlying each Award multiplied by a fraction, the numerator of which is the number of days elapsed from the commencement of the applicable Restriction Period through the date of termination, and the denominator of which is the number of days in such Restriction Period (each a "Retained Restricted Award"), and the remainder of each Award shall be forfeited and canceled as of the date of such termination. The Restriction Period on a Retained Restricted Award shall lapse upon the Participant's termination of employment. Settlement of a Retained Restricted Award shall be made at the time and in the manner provided in Sections 7.3 and 7.4 except that no additional deferrals shall be permitted.
- (b) Any Other Reason. Unless otherwise determined by the Committee at or after the grant date, if a Participant's employment is terminated for any reason during the Restriction Period other than as described in Section 7.5(a), any Restricted Stock and Restricted Stock Units granted to such Participant for which the Restriction Period has not then expired shall be forfeited and canceled as of the date of such termination.

7.6 Deferred Stock Units.

In General. Freestanding Deferred Stock Units may be granted to Participants at such time or times as shall be determined by the Committee without regard to any election by the Participant to defer receipt of any compensation or bonus amount payable to him. The grant date of any freestanding Deferred Stock Units under the Plan will be the date on which such freestanding Deferred Stock Units are awarded by the Committee or on such other future date as the Committee shall determine. In addition, to the extent permitted by applicable law (including section 409A of the Code), on fixed dates established by the Committee and subject to such terms and conditions as the Committee shall determine, the Committee may permit a Participant to elect to defer receipt of all or a portion of his annual compensation and/or incentive bonus ("Deferred Annual Amount") payable by the Company or a Subsidiary and receive in lieu thereof an Award of elective Deferred Stock Units equal to the greatest whole number which may be obtained by dividing (i) the amount of the Deferred Annual Amount by (ii) the Fair Market Value of one share of Common Stock on the date of payment of such compensation and/or annual bonus. Deferred Stock Units shall be evidenced by an Award Agreement that shall specify the number of shares of Common Stock to which the Deferred Stock Units pertains, and such terms and conditions as the Committee shall determine, including customary representations, warranties and covenants with respect to securities law matters. Upon the grant of Deferred Stock Units pursuant to the Plan, the Company shall establish a notional account for the Participant and will record in such account the number of shares of Deferred Stock Units awarded to the Participant. No shares of Common Stock will be

issued to the Participant at the time an award of Deferred Stock Units is granted.

- (b) Rights as a Stockholder. The Committee shall determine whether and to what extent Dividend Equivalents will be credited to the account of, or will be paid currently to, a Participant receiving an Award of Deferred Stock Units. Unless otherwise provided by the Committee at or after the grant date, (i) any cash dividends or distributions credited to the Participant's account shall be deemed to have been invested in additional Deferred Stock Units on the payment date established for the related dividend or distribution in an amount equal to the greatest whole number which may be obtained by dividing (A) the value of such dividend or distribution on the record date by (B) the Fair Market Value of one share of Common Stock on such date, and such additional Deferred Stock Units shall be subject to the same terms and conditions as are applicable in respect of the Deferred Stock Units with respect to which such dividends or distributions were payable, and (ii) if any such dividends or distributions are paid in shares of Common Stock or other securities, such shares and other securities shall be subject to the terms, conditions and restrictions as apply to the Deferred Stock Units with respect to which they were paid. A Participant shall not have any rights as a stockholder in respect of Deferred Stock Units awarded pursuant to the Plan (including, but not limited to, the right to vote on any matter submitted to the Company's stockholders) until such time as the shares of Common Stock attributable to such Deferred Stock Units have been issued to such Participant or his beneficiary.
- (c) Restrictions on Transferability. No Deferred Stock Units may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated. Any attempt by a Participant, directly or indirectly, to offer, transfer, sell, pledge, hypothecate or otherwise dispose of any Deferred Stock Units or any interest therein or any rights relating thereto without complying with the provisions of the Plan shall be void and of no effect.
- Settlement. Unless the Committee determines otherwise at or after the grant date, the Company shall issue the shares of Common Stock underlying any of a Participant's freestanding Deferred Stock Units (and related Dividend Equivalents) for which the Restriction Period shall have lapsed on or prior to the date of such Participant's termination of employment with the Company and any Subsidiary, other than a termination for Cause, as soon as administratively practicable, but not later than 90 days, following the date of such termination of employment (or on such earlier date as the Committee shall permit or such later date as may be elected by the Participant in accordance with section 409A of the Code and the rules and procedures of the Committee or as may be required by applicable law). Unless the Committee determines otherwise at or after the grant date, in the event of the termination of a Participant's employment with the Company and the Subsidiaries for Cause, the Participant shall immediately forfeit all rights with respect to any shares of freestanding Deferred Stock Units (and related Dividend Equivalents) credited to his account, whether or not the Restriction Period shall have then lapsed. Subject to Article VIII and Article X, and the last sentence of Section 7.6(a), unless the Committee determines otherwise at or after the grant date, the Company shall issue the shares of Common Stock underlying any of a Participant's elective Deferred Stock Units (and related Dividend Equivalents) credited to such Participant's account under the Plan as soon as administratively practicable, but not later than 90 days, following the date of such Participant's termination of employment (or such later date as may be elected by the Participant in accordance with the rules and procedures of the Committee or as may be required by applicable law). The Committee may provide in the Award Agreement applicable to any Award of Deferred Stock Units that, in lieu of issuing shares of Common Stock in settlement of any Deferred Stock Units, the Committee may direct the Company to pay to the Participant the Fair Market Value of the shares of Common Stock corresponding to such Deferred Stock Units in cash, or in any combination of shares of Common Stock and cash having an aggregate Fair Market Value equal to such shares of Common Stock.
- (e) Further Deferral Elections. To the extent permitted by applicable law (including section 409A of the Code), upon such terms and conditions as the Committee may establish from time to time, a Participant may elect to further defer receipt of shares of Common Stock issuable in respect of Deferred Stock Units (or an installment of an Award) for a specified period or until a specified event.
- 7.7 Share Awards. Share Awards may be granted to Participants at such time or times as shall be determined by the Committee on such terms and conditions as the Committee may determine in its discretion. Share Awards may be

made as additional compensation for services rendered by a Participant to the Company or any Subsidiary or may be in lieu of cash or other compensation to which the Participant may be entitled from the Company or any Subsidiary.

ARTICLE VIII

CHANGE IN CONTROL

- 8.1 Accelerated Vesting and Payment.
- In General. Unless otherwise determined by the Committee at or after the grant date, or unless the (a) Committee otherwise determines in the manner set forth in Section 8.2, upon the occurrence of a Change in Control, (i) all Options and Stock Appreciation Rights shall become immediately exercisable, (ii) the Restriction Period on all Restricted Stock, Restricted Stock Units and freestanding Deferred Stock Units shall lapse immediately prior to such Change in Control, and (iii) shares of Common Stock underlying Awards of Restricted Stock Units and Deferred Stock Units shall be issued to each Participant then holding such Award immediately prior to such Change in Control; provided, that, at the discretion of the Committee (as constituted immediately prior to the Change in Control), each such Option, Stock Appreciation Right, Restricted Stock Unit and/or Deferred Stock Unit may be canceled in exchange for an amount equal to the product of (A) (I) in the case of Options and Stock Appreciation Rights, the excess, if any, of the product of the Change in Control Price over the exercise price for such Award, and (II) in the case of other such Awards, the Change in Control Price, multiplied by (B) the aggregate number of shares of Common Stock covered by such Award; provided, further, that where the Change in Control does not constitute a "change in control event" as defined under section 409A of the Code, the shares to be issued, or the amount to be paid, for each Award that constitutes deferred compensation subject to section 409A of the Code shall be paid at the time or schedule applicable to such Awards (assuming for these payment purposes (but not the lapsing of the Restriction Period) that no such Change in Control had occurred). Notwithstanding the foregoing, the Committee may, in its discretion, instead terminate any outstanding Options and Stock Appreciation Rights if either (x) the Company provides holders of such Options and Stock Appreciation Rights with reasonable advance notice to exercise their outstanding and unexercised Options and Stock Appreciation Rights or (y) the Committee reasonably determines that the Change in Control Price is equal to or less than the exercise price for such Options and Stock Appreciation Rights.
- (b) Performance Stock, Performance Stock Units and Performance Units. Performance Stock, Performance Stock Units, and Performance Units that are outstanding in the event of a Change in Control shall be treated as provided in the individual Award Agreement governing such Performance Stock, Performance Stock Units, and Performance Units.
- (c) Timing of Payments. Payment of any amounts calculated in accordance with Section 8.1(a) shall be made in cash or, if determined by the Committee (as constituted immediately prior to the Change in Control), in shares of the common stock of the New Employer having an aggregate fair market value equal to such amount and shall be payable in full, as soon as reasonably practicable, but in no event later than 30 days, following the Change in Control (subject to the payment timing restrictions contained in the second proviso of the first sentence of Section 8.1(a)). For purposes hereof, the fair market value of one share of common stock of the New Employer shall be determined by the Committee (as constituted immediately prior to the consummation of the transaction constituting the Change in Control), in good faith.
- 8.2 Alternative Awards. Notwithstanding Section 8.1, unless otherwise determined by the Committee at or after the grant date, no cancellation, termination, acceleration of exercisability or vesting, lapse of any Restriction Period or settlement or other payment shall occur with respect to any outstanding Award (other than an award of Performance Stock, Performance Stock Units, and Performance Units except as provided therein), if the Committee (as constituted immediately prior to the consummation of the transaction constituting the Change in Control) reasonably determines, in good faith, prior to the Change in Control that such outstanding Awards shall be honored or assumed, or new rights substituted therefor (such honored, assumed or substituted Award being hereinafter referred to as an "Alternative Award") by the New Employer, provided, that any Alternative Award must:

- (i) be based on shares of Common Stock that are traded on an established U.S. securities market or another public market determined by the Committee prior to the Change in Control;
- (ii) provide the Participant (or each Participant in a class of Participants) with rights and entitlements substantially equivalent to or better than the rights, terms and conditions applicable under such Award, including, but not limited to, an identical or better exercise or vesting schedule and identical or better timing and methods of payment (including liquidity rights with respect to shares of Common Stock received in settlement of such Award);
- (iii) have substantially equivalent economic value to such Award (determined at the time of the Change in Control);
- (iv) have terms and conditions which provide that in the event that the Participant suffers an involuntary termination without Cause within two years following the Change in Control, any conditions on the Participant's rights under, or any restrictions on transfer or exercisability applicable to, each such Award held by such Participant shall be waived or shall lapse, as the case may be; and
- (v) not result in adverse tax consequences to the Participant under section 409A of the Code.

ARTICLE IX

EFFECTIVE DATE, AMENDMENT, MODIFICATION AND TERMINATION OF PLAN

The Plan was adopted by the Board on April 7, 2010, and approved by the Company's shareholders at the annual meeting of shareholders held on May 21, 2010. The amendment and restatement of the Plan provided herein is subject to shareholder approval at the 2011 shareholder meeting. The Plan shall continue in effect, unless sooner terminated pursuant to this Article X, until the tenth anniversary of the date on which it is adopted by the Board (or if applicable, the 10 year anniversary of the date of the latest shareholder approval of the Plan). The Board or the Committee may at any time terminate or suspend the Plan, and from time to time may amend or modify the Plan; provided, that without the approval by a majority of the votes cast at a meeting of shareholders at which a quorum representing a majority of the shares of the Company entitled to vote generally in the election of directors is present in person or by proxy, no amendment or modification to the Plan may (i) materially increase the benefits accruing to participants under the Plan, (ii) except as otherwise expressly provided in Section 4.4, materially increase the number of shares of Common Stock subject to the Plan or the individual Award limitations specified in Section 4.3, (iii) modify the restrictions provided in Section 4.5 or (iv) materially modify the requirements for participation in the Plan. No amendment, modification, or termination of the Plan shall in any manner adversely affect any Award theretofore granted under the Plan, without the consent of the Participant. Notwithstanding the foregoing, the Board or Committee may take such actions as it deems appropriate to ensure that the Plan and any Awards may comply with any tax, securities or other applicable law. Nothing herein shall restrict the Committee's ability to exercise its discretionary authority as provided in the Plan. Subject to other applicable provisions of the Plan, all Awards made under the Plan prior to such termination of the Plan shall remain in effect until such Awards have been satisfied or terminated in accordance with the Plan and the terms of such Awards. Except as otherwise determined by the Board, termination of the Plan shall not affect the Committee's ability to exercise the powers granted to it hereunder with respect to Awards granted under the Plan prior to the date of such termination. Following a Change in Control, no action shall be taken under the Plan that will cause any Award that has previously been determined to be (or is determined to be) subject to section 409A of the Code to fail to comply in any respect with section 409A of the Code without the written consent of the Participant.

ARTICLE X

MISCELLANEOUS PROVISIONS

- 10.1 Nontransferability of Awards. No Award shall be assignable or transferable except by will or the laws of descent and distribution; provided, that the Committee or the Board may, except as otherwise provided in the Plan, permit (on such terms and conditions as it shall establish) in its sole discretion a Participant to transfer an Award for no consideration to (i) the Participant's child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse, sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, or sister-in-law, including adoptive relationships, any person sharing the Participant's household (other than a tenant or employee), a trust in which these persons have more than fifty percent of the beneficial interest, a foundation in which these persons (or the Participant) control the management of assets, and any other entity in which these persons (or the Participant) own more than fifty percent of the voting interests, or to organizations qualifying as charitable organizations within the meaning of section 501(c)(3) of the Code or (ii) any other person or entity (each of (i) and (ii), upon such permitted transfer, a "Permitted Transferee"). Except to the extent required by law, no Award shall be subject to any lien, obligation or liability of the Participant. All rights with respect to Awards granted to a Participant under the Plan shall be exercisable during the Participant's lifetime only by such Participant or, if applicable, his or her Permitted Transferee(s). The rights of a Permitted Transferee shall be limited to the rights conveyed to such Permitted Transferee, who shall be subject to and bound by the terms of the agreement or agreements between the Participant and the Company.
- Beneficiary Designation. Each Participant under the Plan may, from time to time, name any beneficiary or beneficiaries (who may be named contingently or successively) to whom any benefit under the Plan is to be paid or by whom any right under the Plan is to be exercised in case of his or her death. Each designation will revoke all prior designations by the same Participant, shall be in a form prescribed by the Committee, and will be effective only when filed by the Participant in writing with the Committee during his or her lifetime. In the absence of any such designation, benefits remaining unpaid at the Participant's death shall be paid to or exercised by the Participant's surviving spouse, if any, or otherwise to or by his or her estate.
- No Guarantee of Employment or Participation. Nothing in the Plan or any Award Agreement shall interfere with or limit in any way the right of the Company or any Subsidiary to terminate any Participant's employment at any time, nor to confer upon any Participant any right to continue in the employ of the Company or any Subsidiary (regardless of whether such termination results in (1) the failure of any Award to vest; (2) the forfeiture of any unvested or vested portion of any Award; and/or (3) any other adverse effect on the individual's interests under the Plan). No Employee shall have a right to be selected as a Participant, or, having been so selected, to receive any future Awards.
- 10.4 Tax Withholding. The Company shall have the right and power to deduct from all amounts paid to a Participant in cash or shares (whether under this Plan or otherwise) or to require a Participant to remit to the Company promptly upon notification of the amount due, an amount (which may include shares of Common Stock) to satisfy the minimum federal, state or local or foreign taxes or other obligations required by law to be withheld with respect thereto with respect to any Award under this Plan. In the case of any Award satisfied in the form of shares of Common Stock, no shares of Common Stock shall be issued unless and until arrangements satisfactory to the Committee shall have been made to satisfy the statutory minimum withholding tax obligations applicable with respect to such Award. The Company may defer payments of cash or issuance or delivery of Common Stock until such requirements are satisfied. Without limiting the generality of the foregoing, the Company shall have the right to retain, or the Committee may, subject to such terms and conditions as it may establish from time to time, permit Participants to elect to tender, shares of Common Stock (including shares of Common Stock issuable in respect of an Award) to satisfy, in whole or in part, the amount required to be withheld (provided that such amount shall not be in excess of

the minimum amount required to satisfy the statutory withholding tax obligations).

- 10.5 Compliance with Legal and Exchange Requirements. The Plan, the granting and exercising of Awards thereunder, and any obligations of the Company under the Plan, shall be subject to all applicable federal and state laws, rules, and regulations, and to such approvals by any regulatory or governmental agency as may be required, and to any rules or regulations of any exchange on which the Common Stock is listed. The Company, in its discretion, may postpone the granting, exercising and settlement of Awards, the issuance or delivery of shares of Common Stock under any Award or any other action permitted under the Plan to permit the Company, with reasonable diligence, to complete such stock exchange listing or registration or qualification of such Shares or other required action under any federal or state law, rule or regulation and may require any Participant to make such representations and furnish such information as it may consider appropriate in connection with the issuance or delivery of shares of Common Stock in compliance with applicable laws, rules and regulations. The Company shall not be obligated by virtue of any provision of the Plan to recognize the exercise or settlement of any Award or to otherwise sell or issue shares of Common Stock in violation of any such laws, rules or regulations, and any postponement of the exercise or settlement of any Award under this provision shall not extend the term of such Awards. Neither the Company nor its directors or officers shall have any obligation or liability to a Participant with respect to any Award (or shares of Common Stock issuable thereunder) that shall lapse because of such postponement.
- Indemnification. To the maximum extent provided by law and by the Company's Certificate of Incorporation and/or By-Laws, each person who is or shall have been a member of the Committee or of the Board shall be indemnified and held harmless by the Company against and from any loss, cost, liability or expense that may be imposed upon or reasonably incurred by him or her in connection with or resulting from any claim, action, suit or proceeding to which he or she may be made a party or in which he or she may be involved by reason of any action taken or failure to act under the Plan and against and from any and all amounts paid by him or her in settlement thereof, with the Company's approval, or paid by him or her in satisfaction of any judgment in any such action, suit or proceeding against him or her, provided he or she shall give the Company an opportunity, at its own expense, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf. The foregoing right of indemnification shall not be exclusive and shall be independent of any other rights of indemnification to which such persons may be entitled under the Company's Certificate of Incorporation or By-laws, by contract, as a matter of law, or otherwise.
- 10.7 No Limitation on Compensation. Nothing in the Plan shall be construed to limit the right of the Company to establish other plans or to pay compensation to its employees, in cash or property, in a manner which is not expressly authorized under the Plan.
- 10.8 Deferrals. The Committee may postpone the exercising of Awards, the issuance or delivery of Common Stock under any Award or any action permitted under the Plan to prevent the Company or any Subsidiary from being denied a Federal income tax deduction with respect to any Award other than an ISO or to the extent required or permitted by applicable law.
- 409A Compliance. The Plan is intended to be administered in a manner consistent with the requirements, where applicable, of section 409A of the Code. Where reasonably possible and practicable, the Plan shall be administered in a manner to avoid the imposition on Participants of immediate tax recognition and additional taxes pursuant to such section 409A. Notwithstanding the foregoing, neither the Company nor the Committee shall have any liability to any person in the event such section 409A applies to any such Award in a manner that results in adverse tax consequences for the Participant or any of his beneficiaries or transferees.

Solely for purposes of determining the time and form of payments due under any Award that is considered nonqualified deferred compensation under section 409A of the Code and that is not otherwise exempt from section 409A of the Code, a Participant shall not be deemed to have incurred a termination of employment unless and until he shall incur a "separation from service" within the meaning of section 409A of the Code. Notwithstanding any other

provision in this Plan, if as of Participant's separation from service, the Participant is a "specified employee" as determined by the Company, then to the extent any amount payable under any Award that is considered nonqualified deferred compensation under section 409A of the Code and that is not otherwise exempt from section 409A of the Code, for which payment is triggered by Participant's separation from service (other than on account of death), and that under the terms of the Award would be payable prior to the six-month anniversary of the Participant's separation from service, such payment shall be delayed until the earlier to occur of (a) the six-month anniversary of such separation from service or (b) the date of the Participant's death.

- 10.10 Governing Law. The Plan shall be construed in accordance with and governed by the laws of the State of New Jersey without reference to principles of conflict of laws which would require application of the law of another jurisdiction, except to the extent that the corporate law of the State of New Jersey specifically and mandatorily applies.
- 10.11 Severability; Blue Pencil. In the event that any one or more of the provisions of this Plan shall be or become invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not be affected thereby. If, in the opinion of any court of competent jurisdiction such covenants are not reasonable in any respect, such court shall have the right, power and authority to excise or modify such provision or provisions of these covenants as to the court shall appear not reasonable and to enforce the remainder of these covenants as so amended.
- 10.12 No Impact On Benefits. Except as may otherwise be specifically stated under any employee benefit plan, policy or program, no amount payable in respect of any Award shall be treated as compensation for purposes of calculating a Participant's right under any such plan, policy or program. No amount payable in respect of any Award pursuant to an Award shall be deemed part of a Participant's regular, recurring compensation for purposes of any termination, indemnity or severance pay laws.
- 10.13 No Constraint on Corporate Action. Nothing in this Plan shall be construed (i) to limit, impair or otherwise affect the Company's right or power to make adjustments, reclassifications, reorganizations or changes of its capital or business structure, or to merge or consolidate, or dissolve, liquidate, sell, or transfer all or any part of its business or assets or (ii) to limit the right or power of the Company, or any Subsidiary to take any action which such entity deems to be necessary or appropriate.
- Headings and Captions. The headings and captions herein are provided for reference and convenience only, shall not be considered part of this Plan, and shall not be employed in the construction of this Plan.
- 10.15 No Trust or Fund Created. Neither the Plan nor any Award shall create or be construed to create a trust or separate fund of any kind or a fiduciary relationship between the Company and a grantee or any other person. To the extent that any grantee or other person acquires a right to receive payments from the Company pursuant to an Award, such right shall be no greater than the right of any unsecured general creditor of the Company.
- 10.16 Fractional Shares. No fractional shares of Common Stock shall be issued or delivered pursuant to the Plan or any Award, and the Committee shall determine whether cash, other securities or other property shall be paid or transferred in lieu of any fractional shares, or whether such fractional shares or any rights thereto shall be canceled, terminated or otherwise eliminated.
- 10.17 Code Section 83(b) Elections. The Company, its Affiliates and the Committee have no responsibility for any Participant's election, attempt to elect or failure to elect to include the value of a Restricted Stock Award or other Award subject to section 83 of the Code in the participant's gross income for the year of payment pursuant to section 83(b) of the Code. Any participant who makes an election pursuant to section 83(b) of the Code will promptly provide the Committee with a copy of the election form.

Table of Contents

- 10.18 No Obligation to Exercise Awards; No Right to Notice of Expiration Date. The grant of an Award of an Option, Stock Appreciation Right or Stock Purchase Right will impose no obligation upon the Participant to exercise the Award. The Company, its Affiliates and the Committee have no obligation to inform a Participant of the date on which any Award lapses except in the Award Agreement.
- 10.19 Right to Offset. Notwithstanding any provisions of the Plan to the contrary, and to the extent permitted by applicable law (including section 409A of the Code), the Company may offset any amounts to be paid to a Participant (or, in the event of the Participant's death, to his beneficiary or estate) under the Plan against any amounts that such Participant may owe to the Company or its Affiliates.
- 10.20 Furnishing Information. A Participant will cooperate with the Committee by furnishing any and all information requested by the Committee and take such other actions as may be requested in order to facilitate the administration of the Plan and the payments of benefits hereunder, including but not limited to taking such physical examinations as the Committee may deem necessary when eligibility or entitlement to any compensation or benefit based on Disability is at issue.

EXHIBIT C

EMCORE CORPORATION

2000 EMPLOYEE STOCK PURCHASE PLAN

AMENDED November 9, 2010

ARTICLE I

ESTABLISHMENT

Purpose

The EMCORE Corporation 2000 Employee Stock Purchase Plan (the "Plan") is hereby established by EMCORE Corporation (the "Company"), the purpose of which is to provide a method whereby employees of the Company or any Designated Subsidiary (as defined herein), will have an opportunity to acquire a proprietary interest in the Company through the purchase of shares of Common Stock. The Plan is also established to help promote the overall financial objectives of the Company's stockholders by promoting those persons participating in the Plan to achieve long-term growth in stockholder equity. The Plan is intended to qualify as an "employee stock purchase plan" under Section 423 of the Internal Revenue Code of 1986, as amended (the "Code"). The provisions of the Plan shall be construed so as to extend and limit participation in a manner consistent with the requirements of Section 423 of the Code and the regulations promulgated thereunder.

ARTICLE II

DEFINITIONS

The following words and phrases, as used herein, shall have the meanings indicated unless the context clearly indicates to the contrary:

- 2.01 Account shall mean the bookkeeping account established on behalf of a Participant to which is credited all contributions paid for the purpose of purchasing Common Stock under the Plan, and to which shall be charged all purchases of Common Stock, or withdrawals, pursuant to the Plan. Such Account shall remain unfunded as described in Section 8.11 of the Plan.
- 2.02 Affiliate shall mean, with respect to any Person, any other Person that, directly or indirectly, controls, is controlled by, or is under common control with, such Person. Any "Relative" (for this purpose, "Relative" means a spouse, child, parent, parent of spouse, sibling or grandchild) of an individual shall be deemed to be an Affiliate of such individual for this purpose. Neither the Company nor any Person controlled by the Company shall be deemed to be an Affiliate of any holder of Common Stock.
- 2.03 Agreement shall mean, either individually or collectively, any subscription, enrollment and/or withholding agreement, in the form prescribed by the Committee, entered into pursuant to the Plan between the Company or a Designated Subsidiary and a Participant. Such Agreement shall be an authorization for the Company or a Designated Subsidiary to withhold amounts from such Participant's Compensation, at the Contribution Rate specified in the Agreement, to be applied to purchase Common Stock.

- 2.04 Beneficial Ownership (including correlative terms) shall have the meaning given such term in Rule 13d-3 promulgated under the Exchange Act.
- 2.05 Beneficiary shall mean the person specified by a Participant in his or her most recent written designation that is filed with the Committee to receive any benefits under the Plan in the event of such Participant's death, in accordance with Section 8.01.

- 2.06 Board shall mean the Board of Directors of the Company
- 2.07 Change in Control shall mean the occurrence of any of the following:
- (a) an acquisition in one transaction or a series of related transactions (other than directly from the Company or pursuant to awards granted under the Plan or compensatory options or other similar awards granted by the Company) of any Voting Securities by any Person, immediately after which such Person has Beneficial Ownership of fifty percent (50%) or more of the combined voting power of the Company's then outstanding Voting Securities; provided, however, in determining whether a Change in Control has occurred pursuant to this Section 2.07(a), Voting Securities which are acquired in a Non-Control Acquisition shall not constitute an acquisition that would cause a Change in Control:
- (b) the individuals who, immediately prior to the Effective Date, are members of the Board (the "Incumbent Board"), cease for any reason to constitute at least a majority of the members of the Board; provided, however, that if the election, or nomination for election, by the Company's common stockholders, of any new director was approved by a vote of at least a majority of the Incumbent Board, such new director shall, for purposes of the Plan, be considered as a member of the Incumbent Board; provided further, however, that no individual shall be considered a member of the Incumbent Board if such individual initially assumed office as a result of either an actual or threatened "Election Contest" (as described in Rule 14a-11 promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a Person other than the Board (a "Proxy Contest") including by reason of any agreement intended to avoid or settle any Election Contest or Proxy Contest; or
- (c) the consummation of:
- (1) a merger, consolidation or reorganization involving the Company unless:
- (A) the stockholders of the Company, immediately before such merger, consolidation or reorganization, own, directly or indirectly, immediately following such merger, consolidation or reorganization, more than fifty percent (50%) of the combined voting power of the outstanding voting securities of the corporation resulting from such merger or consolidation or reorganization (the "Surviving Corporation") in substantially the same proportion as their ownership of the Voting Securities immediately before such merger, consolidation or reorganization,
- (B) the individuals who were members of the Incumbent Board immediately prior to the execution of the agreement providing for such merger, consolidation or reorganization constitute at least a majority of the members of the board of directors of the Surviving Corporation, or a corporation Beneficially Owning, directly or indirectly, a majority of the voting securities of the Surviving Corporation, and
- (C) no Person, other than (i) the Company, (ii) any Related Entity (as defined in Section 2.20), (iii) any employee benefit plan (or any trust forming a part thereof) that, immediately prior to such merger, consolidation or reorganization, was maintained by the Company, the Surviving Corporation, or any Related Entity or (iv) any Person who, together with its Affiliates, immediately prior to such merger, consolidation or reorganization had Beneficial Ownership of fifty percent (50%) or more of the then outstanding Voting Securities, owns, together with its Affiliates, Beneficial Ownership of fifty percent (50%) or more of the combined voting power of the Surviving Corporation's then outstanding voting securities
- (a transaction described in clauses (A) through (C) above is referred to herein as a "Non-Control Transaction");
- (2) a complete liquidation or dissolution of the Company; or

Table of Contents

(3) an agreement for the sale or other disposition of all or substantially all of the assets or business of the Company to any Person (other than a transfer to a Related Entity or the distribution to the Company's stockholders of the stock of a Related Entity or any other assets).

Notwithstanding the foregoing, a Change in Control shall not be deemed to occur solely because any Person (the "Subject Person") acquired Beneficial Ownership of fifty percent (50%) or more of the combined voting power of the then outstanding Voting Securities as a result of the acquisition of Voting Securities by the Company which, by reducing the number of Voting Securities then outstanding, increases the proportional number of shares Beneficially Owned by the Subject Persons, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of Voting Securities by the Company, and (1) before such share acquisition by the Company the Subject Person becomes the Beneficial Owner of any new or additional Voting Securities in a related transaction or (2) after such share acquisition by the Company the Subject Person becomes the Beneficial Owner of any new or additional Voting Securities which in either case increases the percentage of the then outstanding Voting Securities Beneficially Owned by the Subject Person, then a Change in Control shall be deemed to occur.

- 2.08 Commission shall mean the Securities and Exchange Commission or any successor entity or agency.
- 2.09 Committee shall mean the Plan Committee of the Board as described in Article VII.
- 2.10 Compensation shall mean, for the relevant period, (a) the total compensation paid in cash to a Participant by the Company and/or a Designated Subsidiary, including salaries, wages, commissions, overtime pay, shift premiums, bonuses, and incentive compensation, plus (b) any pre-tax contributions made by a Participant under Section 401(k) or 125 of the Code. Compensation shall exclude non-cash items, moving or relocation allowances, geographic hardship pay, car allowances, tuition reimbursements, imputed income attributable to cars or life insurance, severance or notice pay, fringe benefits, contributions (except as provided in clause (b) of the immediately preceding sentence) or benefits received under employee benefit or deferred compensation plans or arrangements, income attributable to stock options and similar items.
- 2.11 Common Stock shall mean shares of common stock of the Company, without par value, or the common stock of any successor to the Company, which is designated for the purposes of the Plan.
- 2.12 Contribution Rate shall be that rate of contribution of Compensation to the Plan stated in the Agreement, subject to determination in accordance with Article IV.
- 2.13 Designated Subsidiary shall mean any Subsidiary that has been designated by the Board from time to time in its sole discretion as eligible to participate in the Plan.
- 2.14 Effective Date shall mean April 1, 2000.
- 2.15 Eligible Employee shall mean any individual who is employed on a full-time or part-time basis by the Company or a Designated Subsidiary on an Enrollment Date, except that the Committee in its sole discretion may exclude:
 - (i) employees whose customary employment is not more than 20 hours per week;
 - (ii) employees whose customary employment is for not more than five months in any calendar year; and

(iii)

employees who are considered to be a highly compensated employee of the Company or Designated Subsidiary within the meaning of Section 414(q) of the Code.

As of the Effective Date, and unless and until the Committee determines otherwise, only those employees described in Section 2.15(i) and (ii) are excluded from the class of Eligible Employees.

- 2.16 Enrollment Date shall mean the first day of each Offering Period.
- 2.17 Exchange Act means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated by the Commission thereunder.
- 2.18 Exercise Date shall mean the last day of each Offering Period.
- 2.19 Fair Market Value of a share of Common Stock as of a given date shall mean: (i) if the Common Stock is listed or admitted to trading on an established stock exchange (including, for this purpose, the Nasdaq National Market), the mean of the highest and lowest sale prices for a share of the Common Stock on the composite tape or in Nasdaq National Market trading as reported in The Wall Street Journal (or, if not so reported, such other nationally recognized reporting source as the Committee shall select) for such date, or, if no such prices are reported for such date, the most recent day for which such prices are available shall be used; (ii) if the Common Stock is not then listed or admitted to trading on such a stock exchange, the mean of the closing representative bid and asked prices for the Common Stock on such date as reported by the Nasdaq Small Cap Market or, if not so reported, by the OTC Bulletin Board (or any successor or similar quotation system regularly reporting the market value of the Common Stock in the over-the-counter market), or, if no such prices are reported for such date, the most recent day for which such prices are available shall be used; or (iii) in the event neither of the valuation methods provided for in clauses (i) and (ii) above are practicable, the fair market value of a share of Common Stock determined by such other reasonable valuation method as the Committee shall, in its discretion, select and apply in good faith as of such date.
- 2.20 Non-Control Acquisition shall mean an acquisition by (1) an employee benefit plan (or a trust forming a part thereof) maintained by (x) the Company or (y) any corporation or other Person of which a majority of its voting power or its voting equity securities or equity interest is owned, directly or indirectly, by the Company (a "Related Entity"), (2) the Company or any Related Entity, (3) any of Thomas Russell, The AER Trust 1997, Robert Louis-Dreyfus, Gallium Enterprises, Inc. and Reuben Richards, or (4) any Person in connection with a Non-Control Transaction.
- 2.21 Offering Period shall mean a period as determined by the Committee during which a Participant's Option may be exercised and the accumulated value of the Participant's Account may be applied to purchase Common Stock. Unless otherwise specified by the Committee, the initial Offering Period will begin on February 25th and end on the last Trading Day on or before August 25th of the same calendar year. The 2nd offer period will begin on August 26th and end on the last Trading Day on or before February 24th. The duration of Offering Periods may be changed by the Committee or the Board pursuant to Section 3.06 or 5.04.
- 2.22 Option shall mean the right to purchase the number of shares of Common Stock specified in accordance with the Plan at a price and for a term fixed in accordance with the Plan, and subject to such other limitations and restrictions as may be imposed by the Plan or the Committee in accordance with the Plan.
- 2.23 Option Price shall mean an amount equal to 85% of the Fair Market Value of a share of Common Stock on the Enrollment Date or Exercise Date, whichever is lower.
- 2.24 Participant shall mean an Eligible Employee who satisfies the eligibility conditions of Article III, and to whom an Option has been granted by the Committee under the Plan.
- 2.25 Person shall mean "person" as such term is used for purposes of Section 13(d) or 14(d) of the Exchange Act, including, without limitation, any individual, corporation, limited liability company, partnership, trust, unincorporated organization, government or any agency or political subdivision thereof, or any other entity or any group of Persons.

2.26 Plan Year shall mean the period of twelve (12) or fewer consecutive months commencing on the Effective Date and ending on December 31st of the same calendar year, and the twelve (12) consecutive month period ending the last day of each December of each calendar year thereafter. The Committee may at any time designate another period as the Plan Year.

Table of Contents

- 2.27 Reserves shall mean the number of shares of Common Stock covered by each Option under the Plan that have not yet been exercised and the number of shares of Common Stock that have been authorized for issuance under the Plan but not yet placed under an Option.
- 2.28 Securities Act shall mean the Securities Act of 1933, as amended, and the rules and regulations promulgated by the Commission thereunder.
- 2.29 Subsidiary shall mean any present or future corporation, domestic or foreign, which is or would be a "subsidiary corporation," as defined under Section 424(f) of the Code, of the Company.
- 2.30 Trading Day shall mean a day on which national stock exchanges are open for trading.
- Voting Securities shall mean all outstanding voting securities of the Company entitled to vote generally in the election of the Board.

ARTICLE III

ELIGIBILITY AND PARTICIPATION

3.01 Initial Eligibility

Any individual who is otherwise an Eligible Employee and who is employed with the Company or a Designated Subsidiary on the Effective Date or becomes employed with the Company or a Designated Subsidiary after the Effective Date and is otherwise an Eligible Employee, may participate in the Plan immediately beginning with the first Offering Period that occurs concurrent with or next following either the Effective Date or that individual's initial date of such employment.

3.02 Leave of Absence

For purposes of the Plan, an individual's employment relationship is still considered to be continuing intact while such individual is on sick leave, or other leave of absence approved by the Committee or the Participant's supervisor; provided, however, that if the period of leave of absence exceeds ninety (90) days and the individual's right to reemployment is not guaranteed either by statute or by contract, the employment relationship shall be deemed to have terminated on the ninety-first (91st) day of such leave.

3.03 Eligibility Restrictions

Notwithstanding any provisions of the Plan to the contrary, no employee of the Company or a Designated Subsidiary shall be granted an Option under the Plan:

- (a) if, immediately after the Option is granted, applying the rules under Section 424(d) of the Code to determine Common Stock ownership, such employee would own, immediately after the Option is granted, five percent (5%) or more of the total combined voting power or value of all classes of stock of the Company or any Subsidiary; or
- (b) which permits such employee's rights to purchase stock under the Plan and any other employee stock purchase plans of the Company or any Subsidiary to accrue at a rate that exceeds \$25,000 (or such other amount as may be adjusted from time to time under applicable provisions of the Code or Regs) in Fair Market Value of Common Stock (determined at the time such Option is granted) for each calendar year in which such Option is outstanding.

3.04 Participation

- (a) An Eligible Employee may commence participation by completing an Agreement authorizing payroll deductions and filing it with the payroll office of the Company prior to the applicable Enrollment Date. Such an Eligible Employee is referred to as a Participant.
- (b) Any payroll deductions for a Participant shall commence on the first payroll date following the Enrollment Date and shall end on the last payroll date in the Offering Period to which such authorization is applicable, unless sooner terminated by the Participant as provided in Article VI.

3.05 Option Grant

On the Enrollment Date of each Offering Period, each Participant participating in the Offering Period shall be granted an Option to purchase on the Exercise Date of such Offering Period (at the appropriate Option Price) up to a number of shares of Common Stock as determined by dividing the particular Participant's payroll deductions that have accumulated prior to such Exercise Date and retained in such Participant's Account as of that Exercise Date by the appropriate Option Price. Such purchase of shares of Common Stock shall be subject to the limitations under Sections 3.03 and 3.09. Exercise of the Option shall occur as provided in Section 3.07, unless the Participant has withdrawn as provided in Article VI. The Option shall expire on the last day of the Offering Period. The Committee may determine that there shall be no Options granted under the Plan for any particular Plan Year.

3.06 Offering Period

The Plan shall be implemented by consecutive Offering Periods of Common Stock. Each Agreement shall specify the Offering Period for which the Option is granted, which shall be determined by the Committee in accordance with the Plan. The Committee shall have the authority to change the duration of Offering Periods, including the commencement dates thereof, with respect to future offerings without approval of the Company's stockholders. Under such circumstances, any change to the Offering Periods shall be announced at least ten (10) days prior to the scheduled beginning of the initial Offering Period to be affected. In no event, however, shall an Offering Period extend beyond the period permitted under Section 423(b)(7) of the Code.

3.07 Exercise of Option

Unless a Participant provides written notice to the Company, or withdraws from the Plan as provided in Article VI, his Option for the purchase of shares shall be exercised automatically on the Exercise Date, and the maximum number of full shares subject to the Option shall be purchased for such Participant at the applicable Option Price, using the accumulated payroll deductions in his Account, subject to the limitations under Sections 3.03 and 3.09. No fractional shares shall be purchased. Any payroll deductions accumulated in an Account that are not sufficient to purchase a full share of Common Stock shall be retained in the Account for the subsequent Offering Period, subject to earlier withdrawal by the Participant as provided in Article VI. Any other monies remaining in a Participant's Account after the Exercise Date shall be returned to the Participant or his Beneficiary in cash, without interest. During a Participant's lifetime, such Participant's Option is exercisable only by such Participant.

3.08 Delivery of Stock

(a) As promptly as practical after each Exercise Date on which a purchase of Common Stock occurs, the Company shall arrange the delivery to each Participant, or his Beneficiary, of a certificate representing the shares of Common Stock purchased upon exercise of such Participant's Option, except that the Committee may determine that such shares shall be held for each Participant's benefit by a broker designated by the Committee unless the Participant

has delivered to the Committee a written election that certificates representing such shares be issued to him. Shares of Common Stock issued upon exercise of an Option and delivered to or for the benefit of a Participant or Beneficiary will be registered in the name of such Participant or Beneficiary, as the case may be. Alternatively, at the direction of a Participant through written notice to the Committee at least ten (10) days prior to the applicable Exercise Date, such shares shall be registered in the names of such Participant and one other person as may be designated by the Participant, as joint tenants with rights of survivorship, community property or as tenants by the entirety, to the extent permitted by applicable law.

(b) The Committee may require a Participant or his Beneficiary to give prompt written notice to the Company concerning any disposition of shares of Common Stock received upon the exercise of such Participant's Option within: (i) two (2) years from the date of granting of such Option to such Participant, (ii) one (1) year from the transfer of such shares of Common Stock to such Participant, or (iii) such other period as the Committee may from time to time determine.

3.09 Maximum Number of Shares

In no event shall the number of shares of Common Stock that a Participant may purchase during any one Offering Period under the Plan exceed the number of shares determined by (a) multiplying twenty percent (20%) of the amount of the Participant's Compensation for the payroll period immediately preceding the date he is first granted an Option for such Offering Period by the number of payroll periods from such date to the end of such Offering Period, and (b) dividing that product by 85% of the Fair Market Value of a share of Common Stock on such date.

3.10 Withholding

At the time an Option is exercised, or at the time some or all of the Common Stock that is issued under the Plan is disposed of, the Company may withhold from any Compensation or other amount payable to the applicable Participant, or require such Participant to remit to the Company (or make other arrangements satisfactory to the Company, as determined in the Committee's discretion, regarding payment to the Company of), the amount necessary for the Company to satisfy any Federal, state or local taxes required by law to be withheld with respect to the shares of Common Stock subject to such Option or disposed of, as a condition to delivery of any certificate or certificates for any such shares of Common Stock. Whenever under the Plan payments are to be made in cash, such payments shall be made net of an amount sufficient to satisfy any Federal, state or local tax or withholding obligations with respect to such payments.

ARTICLE IV

PAYROLL DEDUCTIONS

4.01 Contribution Rate

- (a) At the time a Participant files an Agreement with the Committee authorizing payroll deduction, he may elect to have payroll deductions made on each payday during the Offering Period, and such Contribution Rate shall be a minimum of one percent (1%) and a maximum of ten percent (10%) of the Participant's Compensation in effect on each payroll period during the Offering Period, unless the Committee determines otherwise in a manner applicable uniformly to all Participants. The payroll deductions shall only be made in whole percentages of the Participant's Compensation. Participants may not make any separate cash payments outside payroll deductions under the Plan except as otherwise provided in Section 5.04(d) in the event of a Change in Control.
- (b) A Participant may discontinue his participation in the Plan as provided in Article VI, or may elect to decrease the rate of his payroll deductions during the Offering Period by filing a new Agreement with the Committee that authorizes a change in his Contribution Rate. Such election by the Participant to decrease his Contribution Rate shall only be permitted once during each Offering Period. The Committee may, in its discretion, in a fair and equitable manner, limit the number of Participants who change their Contribution Rate during any Offering Period. Any such change in Contribution Rate accepted by the Committee shall be effective with the first full payroll period following ten (10) business days after the Committee's receipt of the new Agreement authorizing the new Contribution Rate, unless the Committee elects to process a change in the Contribution Rate more quickly. A Participant's authorization to change his Contribution Rate shall remain in effect for successive Offering Periods unless terminated as provided in

Article VI.

Table of Contents

(c) Notwithstanding the foregoing provisions of this Section 4.01, the Committee may decrease a Participant's Contribution Rate, but not below zero percent, at any time during an Offering Period to the extent necessary to comply with Section 423(b)(8) of the Code or Section 3.03 of the Plan. To the extent necessary in such case, payroll deductions shall recommence at the rate provided in such Participant's Agreement at the beginning of the first Offering Period that is scheduled to begin in the following Plan Year, unless the Participant withdraws from the Plan in accordance with Article VI.

4.02 Participant Account

All payroll deductions made for a Participant shall be credited to his Account under the Plan.

4.03 Interest

No interest shall accrue on the payroll deductions of a Participant under the Plan. In addition, no interest shall be paid on any and all money that is distributed to a Participant, or his Beneficiary, pursuant to the provisions of Sections 6.01 and/or 6.03.

ARTICLE V

COMMON STOCK

5.01 Shares Provided

- (a) The maximum number of shares of Common Stock that may be issued under the Plan shall be 4,500,000 shares. This number is subject to an adjustment upon any changes in capitalization of the Company as provided in Section 5.04.
- (b) The Committee may determine, in its sole discretion, to include in the number of shares of Common Stock available under the Plan any shares of Common Stock that cease to be subject to an Option or are forfeited or any shares subject to an Option that terminates without issuance of shares of Common Stock actually being made to the Participant.
- (c) If the number of shares of Common Stock that Participants become entitled to purchase under the Plan is greater than the shares of Common Stock offered in a particular Offering Period or remaining available under the Plan, the available shares of Common Stock shall be allocated by the Committee among such Participants in such manner as the Committee determines is fair and equitable.

5.02 Participant Interest

The Participant shall have no interest as a shareholder, including, without limitation, voting or dividend rights, with respect to shares of Common Stock covered by his Option until such Option has been exercised in accordance with the Plan and his Agreement.

5.03 Restriction of Shares Upon Exercise

The Committee may, in its discretion, require as conditions to the exercise of any Option that the shares of Common Stock reserved for issuance upon the exercise of the Option shall have been duly listed upon a stock exchange, and that either:

- (a) a registration statement under the Securities Act with respect to the shares shall be effective, or
- (b) the Participant shall have represented at the time of purchase, in form and substance satisfactory to the Company, that it is his intention to purchase the shares for investment and not for resale or distribution.

5.04 Changes in Capital

- (a) Subject to any required action by the shareholders of the Company, upon changes in the outstanding Common Stock by reason of a stock split, reverse stock split, stock dividend, combination or exchange of shares, merger, recapitalization, consolidation, corporate separation or division of the Company (including, but not limited to, a split-up, spin-off, split-off or distribution to Company stockholders other than a normal cash dividend), reorganization, reclassification, or increase or decrease in the number of shares of capital stock of the Company effected without receipt of full consideration therefor, or any other similar change affecting the Company's capital structure, the Committee shall make appropriate adjustments, in its discretion, to, or substitute, as applicable, the number, class and kind of shares of stock available for Options under the Plan, outstanding Options and the Reserves, the maximum number of shares that a Participant may purchase per Offering Period, the Option Prices of outstanding Options and any other characteristics or terms of the Options or the Plan as the Committee shall determine are necessary or appropriate to reflect equitably the effects of such changes to the Participants; provided, however, that any fractional shares resulting from any such adjustment shall be eliminated by rounding to the next lower whole number of shares with appropriate payment for such fractional shares as shall be reasonably determined by the Committee. Notice of any such adjustment shall be given by the Committee to each Participant whose Option has been adjusted and such adjustment, whether or not such notice has been given, shall be effective and binding for all purposes of the Plan.
- (b) The existence of the Plan and any Options granted hereunder shall not affect in any way the right or power of the Board or the shareholders of the Company to make or authorize any adjustment, recapitalization, reorganization or other change in the Company's capital structure or its business, any merger or consolidation of the Company or a Subsidiary, any issue of debt, preferred or prior preference stock ahead of or affecting Common Stock, the authorization or issuance of additional shares of Common Stock, the dissolution or liquidation of the Company or any Subsidiary, any sale or transfer of all or part of the Company's or a Subsidiary's assets or business or any other corporate act or proceeding.
- (c) The Board may at any time terminate an Offering Period then in progress and provide, in its discretion, that Participants' then outstanding Account balances shall be used to purchase shares pursuant to Article III or returned to the applicable Participants.
- (d) In the event of a Change in Control, the Committee may, in its discretion:
- (i) permit each Participant to make a single sum payment with respect to his outstanding Option before the Exercise Date equal to the amount the Participant would have contributed as determined by the Committee for the payroll periods remaining until the Exercise Date, and provide for termination of the Offering Period then in progress and purchase of shares pursuant to Article III; or
- (ii) provide for payment in cash to each Participant of the amount standing to his Account plus an amount equal to the highest value of the consideration to be received in connection with such transaction for one share of Common Stock, or, if higher, the highest Fair Market Value of the Common Stock during the 30 consecutive Trading Days immediately prior to the closing date or expiration date of such transaction, less the Option Price of the Participant's Option (determined for all purposes of this Section 5.04(d)(ii) using such closing or termination date as the Exercise Date in applying Section 2.23), multiplied by the number of full shares of Common Stock that could have been purchased for such Participant immediately prior to the Change in Control with the amount standing to his Account at the Option Price, and that all Options so paid shall terminate.

ARTICLE VI

WITHDRAWAL

6.01 General

By written notice to the Company, at any time up to 3 days prior to the last day of any particular Offering Period, a Participant may elect to withdraw all of the accumulated payroll deductions in his Account at such time. All of the accumulated payroll deductions credited to such withdrawing Participant's Account shall be paid to such Participant promptly after receipt of his written notice of withdrawal. In addition, upon the Participant's written notice of withdrawal, the Participant's Option for the Offering Period shall be automatically terminated, and no further payroll deductions for the purchase of shares on behalf of such Participant shall be made for such Offering Period. If a Participant withdraws from an Offering Period, payroll deductions shall not resume at the beginning of the succeeding Offering Period unless the Participant delivers to the Committee a new Agreement authorizing payroll deductions.

6.02 Effect on Subsequent Participation

A Participant's withdrawal from an Offering Period shall not have any effect upon his eligibility to participate in any similar plan that may hereafter be adopted by the Company or a Subsidiary or in succeeding Offering Periods that commence after the termination of the Offering Period from which the Participant withdraws.

6.03 Termination of Employment

Upon termination of employment as an Eligible Employee, for any reason, a Participant shall be deemed to have elected to withdraw from the Plan and the payroll deductions credited to such Participant's Account during the Offering Period but not yet used to exercise the Option shall be returned to such Participant, or, in the case of a Participant's death, the payroll deductions credited to such deceased Participant's Account shall be paid to his Beneficiary or Beneficiaries, and the Participant's Option shall be automatically terminated. A transfer of a Participant's employment between or among the Company and any Designated Subsidiary or Designated Subsidiaries shall not be treated as a termination of employment for purposes of the Plan.

ARTICLE VII

ADMINISTRATION

7.01 Generally

The Plan shall be administered by a committee the members of which are appointed by the Board. The Committee shall consist of no fewer than three (3) members. Notwithstanding the foregoing, the Board, in its absolute discretion, may at any time and from time to time exercise any and all rights, duties and responsibilities of the Committee under the Plan, including, but not limited to, establishing procedures to be followed by the Committee, except with respect to any matters which under any applicable law, regulation or rule are required to be determined in the sole discretion of the Committee. If and to the extent that no Committee exists which has the authority to administer the Plan, the functions of the Committee shall be exercised by the Board. In addition, the Board shall have discretionary authority to designate, from time to time, without approval of the Company's stockholders, those Subsidiaries that shall be Designated Subsidiaries, the employees of which are eligible to participate in the Plan.

7.02 Authority of the Committee

The Committee shall have all authority that may be necessary or helpful to enable it to discharge its responsibilities with respect to the Plan. Without limiting the generality of the foregoing sentence or Section 7.01, subject to the express provisions of the Plan, the Committee shall have full and exclusive discretionary authority to interpret and construe any and all provisions of the Plan and any Agreements, determine eligibility to participate in the Plan, adopt rules and regulations for administering the Plan, adjudicate and determine all disputes arising under or in connection with the Plan, determine whether a particular item is included in "Compensation," and make all other determinations deemed necessary or advisable for administering the Plan. Decisions, actions and determinations by the Committee with respect to the Plan or any Agreement shall be final, conclusive and binding on all parties. Except to the extent prohibited by applicable law or the rules of a stock exchange, the Committee may, in its discretion, from time to time, delegate all or any part of its responsibilities and powers under the Plan to any member or members of the management of the Company, and revoke any such delegation.

7.03 Appointment

The Board may from time to time appoint members to the Committee in substitution for or in addition to members previously appointed and may fill vacancies, however caused, on the Committee. The Committee may select one member as its Chair and shall hold its meetings at such times and places as it shall deem advisable. It may also hold telephonic meetings. A majority of its members shall constitute a quorum. All determinations of the Committee shall be made by a majority of its members. The Committee may correct any defect or omission or reconcile any inconsistency in the Plan or any Agreement in the manner and to the extent the Committee determines to be desirable. Any decision or determination reduced to writing and signed by a majority of the members of the Committee shall be as fully effective as if it had been made by a majority vote at a meeting duly called and held. The Committee may appoint a secretary and shall make such rules and regulations for the conduct of its business as it shall deem advisable.

ARTICLE VIII

MISCELLANEOUS

8.01 Designation of Beneficiary

- (a) A Participant may file with the Committee a written designation of a Beneficiary who is to receive any Common Stock and/or cash from the Participant's Account in the event of such Participant's death subsequent to an Exercise Date on which the Option is exercised but prior to delivery to such Participant of such Common Stock and cash. Unless a Participant's written Beneficiary designation states otherwise, the designated Beneficiary shall also be entitled to receive any cash from the Participant's Account in the event of such Participant's death prior to exercise of his Option.
- (b) A Participant's designation of Beneficiary may be changed by the Participant at any time by written notice to the Committee. In the event of the death of a Participant and in the absence of a valid Beneficiary designation under the Plan at the time of such Participant's death, the Company shall deliver the shares and/or cash to which the deceased Participant was entitled under the Plan to the executor or administrator of the estate of such Participant. If no such executor or administrator has been appointed as can be determined by the Committee, the Company shall deliver such shares and/or cash to the spouse or to any one or more dependents or relatives of the Participant, or if no spouse, dependent or relative is known to the Company, then to such other person as the Committee may designate. Any such delivery or payment shall be a complete discharge of the obligations and liabilities of the Company, the Subsidiaries, the Committee and the Board under the Plan.

8.02 Transferability

Neither payroll deductions credited to the Participant's Account nor any rights with regard to the exercise of an Option or to receive Common Stock under the Plan may be assigned, transferred, pledged, or otherwise disposed of in any way other than by will, the laws of descent and distribution, or as provided under Section 8.01. Any such attempt at assignment, transfer, pledge or other disposition shall be without effect, except that the Company may treat such act as an election to withdraw funds from an Offering Period in accordance with Article VI.

8.03 Conditions Upon Issuance of Shares

- (a) If at any time the Committee shall determine, in its discretion, that the listing, registration and/or qualification of shares of Common Stock upon any securities exchange or under any state or Federal law, or the consent or approval of any governmental regulatory body, is necessary or desirable as a condition of, or in connection with, the sale or purchase of shares of Common Stock hereunder, no Option may be exercised or paid in whole or in part unless and until such listing, registration, qualification, consent and/or approval shall have been effected or obtained, or otherwise provided for, free of any conditions not acceptable to the Committee.
- (b) If at any time counsel to the Company shall be of the opinion that any sale or delivery of shares of Common Stock pursuant to an Option is or may be in the circumstances unlawful, contravene the requirements of any stock exchange, or result in the imposition of excise taxes on the Company or any Subsidiary under the statutes, rules or regulations of any applicable jurisdiction, the Company shall have no obligation to make such sale or delivery, or to make any application or to effect or to maintain any qualification or registration under the Securities Act, or otherwise with respect to shares of Common Stock or Options and the right to exercise any Option shall be suspended until, in the opinion of such counsel, such sale or delivery shall be lawful or will not result in the imposition of excise taxes on the Company or any Subsidiary.
- (c) The Committee, in its absolute discretion, may impose such restrictions on the ownership and transferability of the shares of Common Stock purchasable or otherwise receivable by any person under any Option as it deems appropriate. The certificates evidencing such shares may include any legend that the Committee deems appropriate to reflect any such restrictions.

8.04 Participants Bound by Plan

By accepting any benefit under the Plan, each Participant and each person claiming under or through such Participant shall be conclusively deemed to have indicated their acceptance and ratification of, and consent to, all of the terms and conditions of the Plan and any action taken under the Plan by the Committee, the Company or the Board, in any case in accordance with the terms and conditions of the Plan.

8.05 Use of Funds

All payroll deductions received or held by the Company under the Plan may be used by the Company for any corporate purpose, and the Company shall not be obligated to segregate such payroll deductions.

8.06 Amendment or Termination

The Board may terminate, discontinue, amend or suspend the Plan at any time, with or without notice to Participants. No such termination or amendment of the Plan may materially adversely affect the existing rights of any Participant with respect to any outstanding Option previously granted to such Participant, without the consent of such Participant, except for any amendment or termination permitted by Section 5.04. In addition, no amendment of the Plan by the Board shall, without the approval of the shareholders of the Company, (i) increase the maximum number of shares that may be issued under the Plan or that any Participant may purchase under the Plan in any Offering Period, except pursuant to Section 5.04; (ii) change the class of employees eligible to receive Options under the Plan, except as provided by the Board pursuant to the last sentence of Section 7.01; or (iii) change the formula by which the Option Price is determined under the Plan.

8.07 No Employment Rights

The Plan does not, either directly or indirectly, create an independent right for the benefit of any employee or class of employees to purchase any shares of Common Stock under the Plan. In addition, the Plan does not create in any employee or class of employees any right with respect to continuation of employment by the Company or any Subsidiary, and the Plan shall not be deemed to interfere in any way with the Company's or any Subsidiary's employment at will relationship with the employee and/or interfere in any way with the Company's or any Subsidiary's right to terminate, or otherwise modify, an employee's employment at any time or for any or no reason.

8.08 Indemnification

No current or previous member of the Board, or the Committee, nor any officer or employee of the Company acting on behalf of the Board, or the Committee, shall be personally liable for any action, determination, or interpretation taken or made in good faith with respect to the Plan. All such members of the Board or the Committee and each and every officer or employee of the Company acting on their behalf shall, to the extent permitted by law, be fully indemnified and protected by the Company in respect of any such action, determination or interpretation of the Plan. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such individuals may be entitled under the Company's Certificate of Incorporation, or Bylaws, as a matter of law or otherwise.

8.09 Construction of Plan

Whenever the context so requires, the masculine shall include the feminine and neuter, and the singular shall also include the plural, and conversely. The words "Article" and "Section" herein shall refer to provisions of the Plan, unless expressly indicated otherwise.

8.10 Term of Plan

Following the adoption of the Plan by the Board, and approval of the Plan by the shareholders of the Company who are present and represented at a special or annual meeting of the shareholders where a quorum is present, which approval must occur not earlier than one (1) year before, and not later than one (1) year after, the date the Plan is adopted by the Board, the Plan shall become effective on the Effective Date.

8.11 Unfunded Status of Plan

The Plan shall be an unfunded plan. The Committee may authorize the creation of trusts or other arrangements to meet the obligations created under the Plan to deliver Common Stock or make payments, provided that the existence of such trusts or other arrangements is consistent with the unfunded status of the Plan.

8.12 Governing Law

The law of the State of New Jersey will govern all matters relating to the Plan except to the extent such law is superseded by the laws of the United States.

T 1 1		. ~	
Tahl	ല വ	· ('Ot	ntents
I au	-		ItCIItS

Meeting of Shareholders of the Company, to be held at 10:00 a.m. local time on Tuesday, June 14, 2011, at the Langham Hotel, 1401 South Oak Knoll Avenue, Pasadena, California 91106, or at any adjournments or postponements thereof. THIS PROXY, WHEN PROPERLY EXECUTED, WILL BE VOTED IN THE MANNER DIRECTED HEREIN BY THE UNDERSIGNED SHAREHOLDER. IF NO DIRECTION IS MADE, THIS PROXY WILL BE VOTED "FOR" THE ELECTION OF THE DIRECTOR NOMINEES LISTED IN PROPOSAL (1), "FOR" THE RATIFICATION OF KPMG AS EMCORE'S INDEPENDENT AUDITORS IN PROPOSAL (2), "FOR" THE AUTHORIZATION OF A POSSIBLE REVERSE SPUT OF EMCORE'S COMMON STOCK IN PROPOSAL (3), "FOR" THE AMENDMENTS TO EMCORE'S 2010 EQUITY INCENTIVE PLAN IN PROPOSAL (4), "FOR" THE INCREASE IN THE NUMBER OF SHARES OF COMMON STOCK RESERVED FOR ISSUANCE UNDER EMCORE'S 2000 EMPLOYEE STOCK PURCHASE PLAN IN PROPOSAL (5), "FOR" THE ADVISORY VOTE APPROVING EXECUTIVE COMPENSATION IN PROPOSAL (6), FOR "1 YEAR" IN PROPOSAL (7), AND IN THE DISCRETION OF THE PROXIES UPON SUCH OTHER MATTERS AS MAY PROPERLY COME BEFORE THE ANNUAL MEETING. Continued and to be signed on reverse side